

The turbulent bubble break-up cascade. Part 2. Numerical simulations of breaking waves

Wai Hong Ronald Chan¹, Perry L. Johnson^{1,2}, Parviz Moin^{1,†} and Javier Urzay¹

¹Center for Turbulence Research (CTR), Stanford University, Stanford, CA 94305, USA ²The Henry Samueli School of Engineering, University of California, Irvine, Irvine, CA 92697, USA

(Received 17 June 2020; revised 10 September 2020; accepted 27 November 2020)

Breaking waves generate a distribution of bubble sizes that evolves over time. Knowledge of how this distribution evolves is of practical importance for maritime and climate studies. The analytical framework developed in Part 1 (Chan, Johnson & Moin, J. Fluid Mech., vol. 912, 2021, A42) examined how this evolution is governed by the bubble-mass flux from large- to small-bubble sizes which depends on the rate of break-up events and the distribution of child bubble sizes. These statistics are measured in Part 2 as ensemble-averaged functions of time by simulating ensembles of breaking waves, and identifying and tracking individual bubbles and their break-up events. The large-scale break-up dynamics is seen to be statistically unsteady, and two intervals with distinct characteristics were identified. In the first interval, the dissipation rate and bubble-mass flux are quasi-steady, and the theoretical analysis of Part 1 is supported by all observed statistics, including the expected -10/3 power-law exponent for the super-Hinze-scale size distribution. Strong locality is observed in the corresponding bubble-mass flux, supporting the presence of a super-Hinze-scale break-up cascade. In the second interval, the dissipation rate decays, and the bubble-mass flux increases as smalland intermediate-sized bubbles become more populous. This flux remains strongly local with cascade-like behaviour, but the dominant power-law exponent for the size distribution increases to -8/3 as small bubbles are also depleted more quickly. This suggests the emergence of different physical mechanisms during different phases of the breaking-wave evolution, although size-local break-up remains a dominant theme. Parts 1 and 2 present an analytical toolkit for population balance analysis in two-phase flows.

Key words: break-up/coalescence, gas/liquid flow, turbulence simulation

† Email address for correspondence: moin@stanford.edu

1. Introduction

Breaking waves in oceans generate bubbles of a wide range of sizes (Blanchard & Woodcock 1957; Medwin 1970; Johnson & Cooke 1979; Trevorrow, Vagle & Farmer 1994; Melville 1996; Deane 1997; Vagle & Farmer 1998; Deane & Stokes 2002; and others). A key driver in establishing this wide range of bubble sizes is the turbulence that emerges as these waves break (Kanwisher 1963; Kitaigorodskii et al. 1983; Rapp & Melville 1990; Agrawal et al. 1992; Melville 1994, 1996; Terray et al. 1996; Gemmrich & Farmer 2004; Drazen, Melville & Lenain 2008; Deane, Stokes & Callaghan 2016a,b; Mortazavi et al. 2016; and others). A travelling water wave carries gravitational potential energy from the variation of its surface height, as well as kinetic energy in the coherent motion of the water beneath its surface. As these waves break, a significant proportion of these potential and kinetic energies is either expended in entraining air beneath the surface or converted to turbulent kinetic energy. This turbulent kinetic energy is then cascaded from large to small scales, establishing a wide range of characteristic scales of turbulent motion before being dissipated as thermal energy (Richardson 1922; Kolmogorov 1941; Onsager 1945). The resulting turbulence breaks up sufficiently large entrained air cavities into bubbles of various sizes (Kolmogorov 1949; Hinze 1955). The smaller the size of the bubbles produced by these break-up events, the slower their rise velocity and the longer their residence time in the ocean (Garrettson 1973; Thorpe 1982, 1992; Trevorrow et al. 1994). As they rise towards the ocean surface, these bubbles are known to scavenge surfactants and other microparticles. This delays their coalescence with the atmosphere and with one another, and stabilizes them for a longer period of time in the ocean (Fox & Herzfeld 1954; Turner 1961; Blanchard & Syzdek 1970; Johnson & Cooke 1980; Weber, Blanchard & Syzdek 1983; Johnson & Wangersky 1987; Stefan & Szeri 1999; Takagi & Matsumoto 2011; Czerski 2017; and references therein). Some of these bubbles eventually burst at the surface to form film and jet drops (Blanchard & Woodcock 1957; Thorpe 1992; Deane & Stokes 2002; Veron 2015; and references therein). On the whole, then, these bubbles impart an enduring influence on various physical phenomena.

Knowledge of the distribution of bubble sizes informs the total interfacial area, as well as size-dependent effects. These are important for quantifying bubble- and drop-mediated mass, momentum and energy transport near the wave surface (Kanwisher 1963; Atkinson 1973; Thorpe 1982, 1992, 1995; Woolf 1993; Melville 1996; Wanninkhof et al. 2009; Emerson & Bushinsky 2016; and references therein), the reflection and scattering of solar and other electromagnetic radiation by the water-wave surface and surrounding air-water interfaces (Stramski 1994; Zhang, Lewis & Johnson 1998; Stefan & Szeri 1999; Terrill, Melville & Stramski 2001; Reed & Milgram 2002; Stramski et al. 2004; Zhang et al. 2004; Seitz 2011; Twardowski et al. 2012; Crook, Jackson & Forster 2016; and references therein) as well as the generation, scattering and propagation of acoustic waves beneath the water-wave surface (Strasberg 1956; Schulkin 1968, 1969; Medwin 1970; Farmer & Lemon 1984; Prosperetti 1988; Hall 1989; Melville 1996; Deane 1997, 2016; Duineveld 1998; Vagle & Farmer 1998; Farmer, Deane & Vagle 2001; Stanic et al. 2009; Czerski & Deane 2010; and references therein). In order to rigorously quantify these effects, one needs to have a good grasp of the initial distribution of bubble sizes generated in the wave-breaking process, as well as how and why this distribution evolves while the resulting turbulence dissipates and the bubbles rise to the surface.

A number of experiments have measured the bubble-size distribution due to breaking waves generated under a variety of conditions. Among these, the experiments by Deane & Stokes (2002) and Blenkinsopp & Chaplin (2010) were able to resolve bubbles of sizes spanning over two decades (from under 100 μ m to over 10 mm) – a formidable size

range for a single measurement set-up. Crucially, these experiments could characterize the size distribution early in the wave-breaking process, as the optical methods employed are suitable for these high-void-fraction scenarios. While there are differences in the reported distributions due to the different wave parameters and measurement techniques, as well as the precise stages of wave breaking that were characterized, there appear to be two common themes: first, the distribution of bubble sizes has distinct scalings in different bubble-size subranges; second, the distribution evolves noticeably in time as the wave breaks. As alluded to in Part 1 (Chan, Johnson & Moin 2021), these observations imply that different physical mechanisms are at play at different length and time scales in the formation and dynamics of bubbles in these waves. Similar observations were made in the three-dimensional breaking-wave simulations of Wang, Yang & Stern (2016) and Deike, Melville & Popinet (2016), which draw heavily on the foundational two-dimensional simulations of Chen et al. (1999) and Iafrati (2009, 2011). Numerical simulations have fewer limitations on the access to detailed spatial information than experimental studies, which is crucial for characterizing these physical mechanisms. However, the separation of scales inherent in these oceanic systems makes it prohibitively costly to generate large simulation ensembles that resolve the range of bubble sizes accessed in these seminal experiments. A large number of ensemble realizations are necessary to achieve statistical convergence for various bubble statistics in these statistically unsteady waves. As such, mechanisms governing bubble formation and dynamics have not all been straightforward to isolate and remain a subject of active research. A number of candidate mechanisms have been proposed for various bubble-size subranges and wave-breaking stages, including bubble break-up by turbulence (Kolmogorov 1949; Hinze 1955), air entrainment and microbubble formation due to plunging jets and drops (Deane & Stokes 2002; Kiger & Duncan 2012; Chan, Urzay & Moin 2018c; Mirjalili, Chan & Mani 2018; Chan et al. 2019; Mirjalili & Mani 2020), buoyant degassing and dissolution (Garrett, Li & Farmer 2000) and intermittency in the energy dissipation rate (Garrett et al. 2000; Gemmrich 2010), although this intermittency remains a topic of active investigation (Deane 2016; Deane et al. 2016b). Note that microbubble formation due to plunging jets and drops may entail the direct generation of sub-Hinze-scale bubbles with subunity Weber numbers from larger air sheets and films, which may bypass the action of turbulent break-up (Deane & Stokes 2002: Chan et al. 2018c, 2019: Mirialili et al. 2018: Mirialili & Mani 2020). As is discussed in §3, the simulations of this work do not have numerical support for sub-Hinze-scale bubbles, and thus these bubbles are out of the scope of the current work.

Break-up by turbulence is often cited as the dominant mechanism for the generation of bubbles of intermediate sizes during the active wave-breaking phase, when air is entrained beneath the wave surface. The sizes of these fragmenting bubbles are typically comparable to or larger than the global Hinze scale, at which forces of capillary and inertial origins are approximately in balance (Kolmogorov 1949; Hinze 1955). This characteristic length scale was introduced in Part 1 and is dimensionally about a millimetre in most terrestrial oceanic waves (Deane *et al.* 2016*a*). Because these bubbles are smaller than the integral scales of the system, the large-scale flow geometry should not have a strong influence on the break-up dynamics. Garrett *et al.* (2000) suggested that the break-up of these bubbles by turbulence occurs through a quasi-steady cascade of bubble mass from large- to small-bubble sizes. (This quasi-steadiness is from the point of view of the small- and intermediate-sized bubbles.) As discussed in the introduction of Part 1, prior studies have theorized and observed in various bubbly flows that this mechanism implies a $D^{-2/3}$ power-law scaling for the break-up frequency of bubbles of size D (Hinze 1955; Martínez-Bazán, Montañés & Lasheras 1999; Rodríguez-Rodríguez, Gordillo & Martínez-Bazán 2006; Chan *et al.* 2018*b*), and a $D^{-10/3}$ power-law scaling for the corresponding bubble-size distribution (Filippov 1961; Garrett *et al.* 2000; Deane & Stokes 2002; Mortazavi 2016; Deike *et al.* 2016; Wang *et al.* 2016; Chan *et al.* 2018*b,c*). The mathematical formalism in Part 1, which quantifies the average rate of bubble-mass transfer from large- to small-bubble sizes due to break-up events, further demonstrates that these power-law scalings are directly compatible with the notions of locality and self-similarity in the bubble-mass-transfer process. This compatibility confirms key physical aspects of the bubble break-up cascade phenomenology, and provides a theoretical basis for the dimensional analysis of Garrett *et al.* (2000).

This paper (Part 2) aims to determine the extent to which the theoretical results of Part 1 are supported by numerical simulations, by algorithmically embedding this analytical toolkit and directly measuring its key component metrics in the simulations. While previous breaking-wave simulations have observed a -10/3 power-law exponent in the bubble-size distribution, this work seeks a more direct observation of the bubble break-up cascade. The bubble-mass-transfer rate introduced in Part 1 achieves this objective, and is itself dependent on the size distribution, the break-up frequency as well as the distribution of child bubble sizes. In Part 2, these bubble statistics are measured by averaging over ensembles of breaking-wave simulations. They were computed using novel post-processing algorithms that identify and track individual bubbles, and record the details of individual break-up events. The evolution of these ensemble-averaged statistics with time is studied, building on the work of Deike et al. (2016), and the effect of time averaging is elucidated with reference to the time-averaged statistics of Deane & Stokes (2002), Wang et al. (2016) and Deike et al. (2016). The statistics discussed in Part 2 provide direct support for the existence of the quasi-steady bubble break-up cascade proposed by Garrett et al. (2000) and theoretically examined in Part 1, at least during the early wave-breaking stages. This support lends credence to the usage of this analytical toolkit for examining the break-up dynamics of turbulent two-phase flows in general, and provides an avenue for developing and validating break-up kernels typically used for population balance modelling.

The dynamics of bubble break-up appears to be distinct in the early and late wave-breaking stages, suggesting the emergence of distinct bubble generation and evolution mechanisms. The size distribution was observed to deviate from the aforementioned $D^{-10/3}$ power-law scaling by Deane & Stokes (2002), Tavakolinejad (2010) and Masnadi *et al.* (2020) late in the wave-breaking process. Prior breaking-wave simulations either did not investigate the evolution of the ensemble-averaged size distribution as a function of time, or did not conclusively recover these alternative scalings in their ensemble-averaged size distribution in the late wave-breaking stages. The results of this work indicate that the size distribution and other bubble statistics are indeed evolving functions of bubble size and time. This paper systematically identifies bubble-size subranges and times over which the $D^{-10/3}$ scaling is not fully recovered in the size distribution. The characteristics of the bubble-mass-transfer rate are examined in these bubble-size subranges and times in order to explore candidate mechanisms for the formation and dynamics of bubbles under these conditions.

This paper is organized as follows. In § 2, the key results of Part 1 are recapitulated and reformulated with the inclusion of volume averaging for the ensembles of breaking-wave simulations described in § 3. Features of the algorithms used to identify individual bubbles in these ensembles, and to detect break-up and coalescence events by tracking the lineages of these bubbles, are briefly discussed in § 4. The resulting bubble statistics are examined in § 5 in relation to the theoretical analysis of Part 1 and § 2. Finally, conclusions are drawn in § 6.

https://doi.org/10.1017/jfm.2020.1084 Published online by Cambridge University Press

2. Analysis of the bubble break-up cascade for breaking-wave simulations

Volume averaging $(\bar{\})$ is commonly used in the computation of bubble statistics in both numerical and experimental studies of breaking waves. As such, this section recasts the essential expressions of the mathematical formulation for bubble statistics in Part 1 using volume averaging in order to facilitate the subsequent analysis of bubble break-up in breaking-wave simulations. References to §§ 4 and 5 are also made to indicate how these statistics are computed in the simulations and where they are reported, respectively. The reader is referred to Part 1 for a more detailed development and interpretation of the analysis tools summarized here, as well as references to relevant prior art.

Volume averaging may be interpreted as an averaging procedure over the small, localized regions in which the turbulent energy and bubble-mass cascades are postulated to coexist, as reviewed in § 2 of Part 1. The volume-averaging procedure is further discussed in appendix A. The order of magnitude of the correspondingly averaged dissipation rate $\bar{\varepsilon}$ may be estimated in a global sense by u_L^3/L , where L is the wavelength of the dominant wave, $u_L = (gL)^{1/2}/(2\pi)^{1/2}$ is the corresponding wave phase velocity and g is the magnitude of standard gravity. This estimate for the characteristic dissipation rate is addressed in more detail in appendix B. It corresponds to the following dimensional expressions for the global Kolmogorov and Hinze scales:

$$L_K \sim \left(\frac{\mu_l}{\rho_l}\right)^{3/4} (\bar{\varepsilon})^{-1/4}, \qquad (2.1)$$

$$L_H \sim \left(\frac{\sigma}{\rho_l}\right)^{3/5} (\bar{\varepsilon})^{-2/5} , \qquad (2.2)$$

where ρ_l and μ_l denote the density and dynamic viscosity of the liquid phase, respectively, and σ denotes the air–water surface tension coefficient.

2.1. The bubble-mass-transfer flux

The population balance equation describing the time evolution of the volume-weighted size distribution, fD^3 , was discussed in § 3.2 of Part 1. Volume averaging the phase-space-based equation (3.4) in Part 1 yields

$$\frac{\partial [\bar{f}(D;t)D^3]}{\partial t} + \frac{\partial [\bar{v}_D \bar{f}(D;t)D^3]}{\partial D} = \bar{H}(D;t).$$
(2.3)

Correspondingly, volume averaging the kernel-based equation (3.7) in Part 1 yields

$$\frac{\partial [\overline{f}(D;t)D^3]}{\partial t} = \overline{T_b}(D;t) + \overline{T_c}(D;t).$$
(2.4)

As with the derivation of (3.10) in Part 1, these two equations may be compared and reduced by assuming that size-local break-up dominates in an intermediate subrange of bubble sizes to yield a simplified evolution equation for $\bar{f}D^3$:

$$\frac{\partial [\bar{f}(D;t)D^3]}{\partial t} = -\frac{\partial [\overline{v_D f}(D;t)D^3]}{\partial D} = \overline{T_b}(D;t).$$
(2.5)

The size distribution, \overline{f} , is computed using the identification algorithm in § 4.1, and is further discussed in § 5.1. Note that the atmosphere above the wave is treated as a large gaseous reservoir in this work. As such, entrainment events are not included in the tally of

break-up events, and degassing events are not included in the tally of coalescence events. The local break-up flux $\overline{W_b} = -\overline{v_{Df}}D^3$ may be interpreted as the ensemble-averaged and volume-averaged rate of bubble-mass transfer from all bubble sizes larger than D to all bubble sizes smaller than D. One may derive an expression for the analogous flux arising from $\overline{T_b}$, which is obtained by volume averaging (3.11) in Part 1 as follows:

$$\overline{T_b}(D;t) = \int_D^\infty \mathrm{d}D_p \check{q}_b\left(D;t|D_p\right) \overline{g_{bf}}\left(D_p;t\right) D^3 - \overline{g_{bf}}(D;t) D^3.$$
(2.6)

The ensemble-averaged and volume-averaged differential break-up rate $\overline{g_b f}(D_p; t)$ is the expected number of break-up events per unit time, unit domain volume and unit size for bubbles of size D_p at time t, including all events throughout the characteristic wave volume L^3 of each ensemble realization. Then, $\check{q}_b(D_c; t|D_p)$ describes the probability distribution of sizes of child bubbles in these events. Note that each relevant break-up event is weighted equally within each ensemble realization in the computation of \check{q}_b . As such, \check{q}_b satisfies the same constraints as the analogous distribution q_b in Part 1. The quantities $\overline{g_b f}$ and \check{q}_b are both computed using the tracking algorithm in § 4.2, and are further discussed in §§ 5.2 and 5.3, respectively. The corresponding bubble-mass-transfer flux, $\overline{W_b}$, may then be expressed as

$$\overline{W_b}(D;t) = \int_0^D \mathrm{d}D_c D_c^3 \int_D^\infty \mathrm{d}D_p \check{q}_b \left(D_c;t | D_p \right) \overline{g_{bf}}(D_p;t).$$
(2.7)

The transfer flux $\overline{W_b}$ may also be directly computed using the tracking algorithm in § 4.2, and is further discussed in § 5.4. In particular, its variations with bubble size and time are discussed in § 5.4.2. In a system where entrainment at large sizes and break-up towards smaller sizes are the dominant physical mechanisms present, the break-up flux $\overline{W_b}$ may be used as a proxy for the entrainment flux, as suggested in figure 9 of Part 1. Note that a similar procedure may be used to derive the bubble coalescence flux $\overline{W_c}$ from an appropriate model coalescence kernel $\overline{T_c}$.

2.2. Assessing locality

The expressions (3.15) and (3.16) derived in Part 1 to quantify infrared and ultraviolet locality may be rewritten as

$$\overline{I_p}(D_p; t|D) = \int_0^D \mathrm{d}D_c D_c^3 \check{q}_b \left(D_c; t|D_p \right) \overline{g_{bf}} \left(D_p; t \right), \qquad (2.8)$$

$$\overline{I_c}(D_c;t|D) = \int_D^\infty \mathrm{d}D_p D_c^3 \check{q}_b \left(D_c;t|D_p\right) \overline{g_{bf}} \left(D_p;t\right).$$
(2.9)

These quantities may also be directly computed using the tracking algorithm in § 4.2, and are further discussed in § 5.4.1.

In this work, the infrared and ultraviolet locality of $\overline{W_b}$ are investigated using ensembles of breaking-wave simulations in two ways. First, the individual scalings of $\check{q}_b(D_c; t|D_p)$ and $\overline{g_bf}(D_p; t)$ with D_c and D_p are computed and compared against the scalings derived in Part 1. Second, $\overline{I_p}$ and $\overline{I_c}$ are directly computed by summing the mass transfers from all relevant break-up events in the simulations, as outlined in appendix B of Part 1. The decay rates of $\overline{I_p}(D_p; t|D)$ and $\overline{I_c}(D_c; t|D)$ with increasing D_p and decreasing D_c , respectively, are then examined. Before these statistics are discussed in § 5, the breaking-wave simulation ensembles are first described in § 3, and the algorithms used to obtain these statistics are introduced in § 4.

3. The breaking-wave simulation ensembles

3.1. Flow solver

In order to investigate the evolution of the bubble-size distribution \bar{f} , as well as the other bubble statistics introduced in § 2, ensembles of numerical simulations of breaking waves were generated using an unstructured, collocated, node-centred and unsplit geometric volume-of-fluid-based flow solver developed for incompressible and immiscible two-phase flows (Ham et al. 2014; Kim et al. 2014; Bravo et al. 2018b). Among other flow variables, the solver tracks the spatially discretized volume fraction of one of the two phases ϕ , noting that the local sum of the volume fractions of the two phases is 1 by definition. Without loss of generality, it is assumed that the solver is used to simulate a liquid-gas mixture, and ϕ denotes the local volume fraction of the liquid phase. The mixture is assumed to be non-reacting and electrically uncharged, and no mass transfer takes place between the phases. In each interfacial computational median dual cell, the interface is represented using the conventional piecewise-linear interface calculation scheme, while the corresponding interface normal vector is computed via a reconstructed distance function (Cummins, Francois & Kothe 2005), and the corresponding interface curvature is estimated using the second-order direct front curvature method (Herrmann 2008). Note that the piecewise-linear interface calculation scheme is known to minimize jetsam and flotsam, and thus suppresses spurious bubble break-up, compared to the traditional simple line interface calculation scheme with which these structures are commonly associated (Scardovelli & Zaleski 1999). (See also a related discussion, and a more detailed definition of jetsam and flotsam, in § 4.1.) The density and viscosity of the fluid in these interfacial cells are defined as

$$\rho = \phi \rho_l + (1 - \phi) \rho_g, \tag{3.1}$$

$$\mu = \phi \mu_l + (1 - \phi) \mu_g, \tag{3.2}$$

where ρ_g and μ_g denote the density and dynamic viscosity of the gaseous phase, respectively, and $0 < \phi < 1$. Mass and momentum are consistently advected at the interface (Mirjalili, Ivey & Mani 2019) using a variant of the non-intersecting flux polyhedron advection algorithm (Ivey & Moin 2017) to maintain numerical stability for large-density-ratio flows, while the viscous terms in the momentum equation are implicitly time-advanced with the second-order Crank–Nicolson scheme. The fractional-step method is used in order to include a pressure-projection step to maintain a divergence-free velocity field. Within this method, the surface tension force is treated using a balanced-force algorithm (Francois *et al.* 2006; Herrmann 2008) involving the usual continuum-surface-force representation to minimize spurious currents of capillary origin. Surface tension gradients are assumed to be negligible. The solver has been used to simulate a number of liquid jet break-up problems involving primary atomization, including laminar, transitional and turbulent jets in quiescent gas (Bravo *et al.* 2015, 2016, 2018*b*, 2019), jets in cross-flows (Bravo *et al.* 2018*a*) as well as jets from swirling injectors (Kim *et al.* 2014; Ham *et al.* 2014).

3.2. Description of set-up of ensembles

In this work, a baseline ensemble of 30 numerical simulations of breaking third-order Stokes water waves in air was generated. Specifically, the density and viscosity ratios of the

Quantity	Dimensional notation	Non-dimensional notation	
Lengths/coordinates	$\hat{x}, \hat{y}, \hat{z}, \hat{x}, \hat{\eta}$	$x, y, z, x, \eta = \{\hat{x}, \hat{y}, \hat{z}, \hat{x}, \hat{\eta}\}/L$	
Bubble size	D	D/L	
Gradient operator	$\hat{ abla}$	$ abla = L \hat{ abla}$	
Velocity components	$\hat{u}, \hat{v}, \hat{w}$	$u, v, w = {\hat{u}, \hat{v}, \hat{w}} / \sqrt{(gL)/(2\pi)} = {\hat{u}, \hat{v}, \hat{w}} / u_L$	
Time	î	$t = \hat{t} / \sqrt{L/g}$	
Energies	$\hat{E_k}, \hat{E_p}, \hat{E_s}, \hat{E_t}$	E_k, E_p, E_s, E_t	

Table 1. Characteristic scales used for non-dimensionalization. The characteristic time scale was selected to be equal to that used by Wang *et al.* (2016). With this time scale, the first wave-surface impact after the wave overturns occurs shortly after t = 1, while a single wave period corresponds to 2.5 characteristic times. The energies are defined in § 3.3.

two phases are equal to those of an air-water mixture. These waves have the dimensionless integral-scale parameters $We_L = 1.6 \times 10^3$ and $Re_L = 1.8 \times 10^5$, which match those of a 27 cm long water wave at atmospheric conditions, and are similar to those selected by Wang *et al.* (2016). Here, Re_L is the integral-scale Reynolds number $Re_L = \rho_l u_L L/\mu_l$ and We_L is the integral-scale Weber number $We_L = \rho_l u_L^2 L/\sigma$. The initial conditions employed are similar to those adopted by Chen *et al.* (1999), Iafrati (2009, 2011) and Wang *et al.* (2016): the initial dimensionless wave-surface height η was initialized in terms of the non-dimensional streamwise coordinate x using

$$\eta(x,t=0) = \frac{1}{2\pi} \left[S_1 \cos(2\pi x) + \frac{1}{2} S_1^2 \cos(4\pi x) + \frac{3}{8} S_1^3 \cos(6\pi x) \right], \quad (3.3)$$

where $S_1 = a_1k_1 = 0.55$ is the slope of the fundamental wave component, a_1 and $k_1 = 2\pi/L$ are, respectively, the corresponding dimensional amplitude and wavenumber and the wave propagates in the *x* direction. Note that the lengths in the expression for η have been non-dimensionalized by the wavelength of the fundamental wave component, *L*. A list of characteristic scales used for non-dimensionalization is provided in table 1.

In order to generate an ensemble of statistically independent but similar realizations (i.e. with the same configuration), the free surface was further perturbed by a set of random displacements $\Delta \eta = \Delta \eta(z)$ smaller than the local grid spacing, such that every interfacial mesh node with the same *z* (spanwise) coordinate within the same realization has the same $\Delta \eta$. While the shift $\Delta \eta$ is not explicitly resolved by the mesh, it perturbs the volume fraction in these interfacial nodes and provides an implicit disturbance to the original interface. This choice of perturbation preserves the modal content of the wave profile in the streamwise direction since the mesh employed in this work is Cartesian. In other words, the initial conditions are effectively two-dimensional except for a perturbation of spanwise modes. As in the works cited above, the air above the wave surface was initialized at rest, while the water below the surface was initialized with the following dimensionless velocity field (Iafrati 2009, 2011):

$$u(x, y, t = 0) = S_1 \sqrt{1 + S_1^2} \cos(2\pi x) \exp(2\pi y), \qquad (3.4)$$

$$v(x, y, t = 0) = S_1 \sqrt{1 + S_1^2 \sin(2\pi x)} \exp(2\pi y),$$
 (3.5)

912 A43-8



Figure 1. Snapshots of the spanwise cross-section of the $\phi = 0.5$ isosurface of one of the baseline ensemble realizations corresponding to (a) the initial waveform (t = 0) and (b) the wave sometime after breaking has occurred (t = 4.03). The grey lines in the background depict the local mesh configuration. The nodal mesh volumes in the uniform-resolution regions are isotropic.

where y is the non-dimensional coordinate antiparallel to gravity. Periodic boundary conditions were employed in the x and z directions, while free-slip boundary conditions were employed on the two remaining boundary faces of the computational domain, which is a cube of length L. In each of the realizations in this ensemble, the computational mesh consists of about 4.2 million mesh nodes with a non-dimensional minimum grid spacing of 1/216. This is equivalent to a dimensional grid spacing of 1.25 mm for a water wave in air at atmospheric conditions with the aforementioned dimensionless parameters. To put this in context, the dimensional Hinze scale (2.2) takes a value of the order of 3 mm. As evidenced in figures 3, 4, 6 and 7, mesh insensitivity is observed in the energetics and bubble statistics at this mesh resolution. For the latter, this insensitivity is observed over a subrange of super-Hinze-scale bubble sizes where turbulent break-up is expected to be dominant. (See also the description of a more resolved ensemble used in this mesh sensitivity study in the ensuing paragraph.) This resolution was selected to permit more ensemble realizations and enable statistical convergence in a time-resolved sense. The mesh is non-uniform and is finer closer to the central region of the domain, such that a large majority of the generated bubbles are resolved using the minimum grid spacing. Snapshots of the initial and post-breaking waveforms, with the computational mesh overlaid, are illustrated for one of the realizations in figure 1. The non-dimensional time step adopted is $\Delta t = 6.0 \times 10^{-5}$, leading to a maximum Courant number of about 0.1 throughout the computational domain in the course of the simulations. The relatively small Courant number reduces the shape mismatch between the streak tube and the numerical flux polyhedron in the volume-of-fluid-based advection scheme (Ivey & Moin 2017), which is important to manage in the case of inadequately resolved mixed-phase regions. The evolution of the waveform for one of these realizations is depicted from two different viewpoints in figure 2 to illustrate the interfacial features generated as the wave breaks.

Besides the baseline case, an ensemble of three numerical simulations with the same wave parameters and a higher mesh resolution was also generated for a mesh convergence study of the specifically desired quantities studied in this work, such as the bubble-size distribution. In this ensemble, the computational mesh consists of about 32 million mesh nodes with a non-dimensional minimum grid spacing of 1/432 (dimensionally equivalent



Figure 2. Snapshots of (a,c,e,g) an axonometric projection from above the wave and (b,d,f,h) the spanwise cross-section of the $\phi = 0.5$ isosurface of one of the baseline ensemble realizations corresponding to various times in the wave-breaking process: $(a,b) \ t = 2.00$, $(c,d) \ t = 3.01$, $(e,f) \ t = 4.03$ and $(g,h) \ t = 5.04$. The snapshots are sampled at an interval of 1.01 characteristic times. The waves are travelling from left to right, and wrap around the domain due to the periodic boundary conditions in the streamwise direction.

to 0.63 mm) and a non-dimensional time step of 1.2×10^{-5} . A more detailed rendering of one of the realizations from this ensemble may be found in the video discussed by Chan *et al.* (2019). Note that even with this higher spatial resolution, the Kolmogorov length scale (2.1), $L_K \sim 30 \,\mu$ m, remains inaccessible as in many previous breaking-wave simulations. In fact, it may be shown that the smallest scales of turbulent motion near a phase interface could require sub-Kolmogorov resolution (Dodd & Jofre 2019) that is rarely attained. As remarked in the introduction, an ensemble of direct numerical simulations of breaking waves that resolves these dynamics with converged statistics remains a formidable undertaking due to the inherent scale separation in these oceanic systems, and in energetic multiphase flows in general. In view of these limitations, only intermediate-scale quantities where sub-Hinze-scale dynamics has a limited influence are discussed in this work, including the bubble statistics introduced in § 2. In other words, the formation of sub-Hinze-scale bubbles, which may be due to mechanisms distinct from a turbulent break-up cascade, is beyond the scope of this work.

3.3. Time evolution of the breaking wave

From the snapshots of the plunging wave-breaking process in figure 2, it is evident that the wave profile and accompanying distribution of bubbles dramatically evolve in the span of

Bubble statistics from breaking-wave simulations

about 1 to 2 wave periods, or 3 to 5 characteristic times. A qualitatively similar evolution was observed in the studies by Bonmarin (1989), Chen *et al.* (1999), Deane & Stokes (2002), Blenkinsopp & Chaplin (2007), Blenkinsopp & Chaplin (2010), Iafrati (2009), Rojas & Loewen (2010), Kiger & Duncan (2012), Deike, Popinet & Melville (2015), Deike *et al.* (2016), Lim *et al.* (2015) and Wang *et al.* (2016). In particular, a number of tubular air cavities are formed, and then subsequently deformed and ruptured beneath multiple surface splash-ups. This bubble fragmentation eventually ceases as large bubbles degas more quickly, leaving a plume of smaller, slowly rising bubbles. More details are provided by Chan (2020).

This evolution in the wave dynamics is also manifested in the energetics of the wave, whose time evolution averaged over the wave ensemble is plotted in figure 3. The volume-integrated total energy plotted in figure 3(a) is defined as follows:

$$E_{t} = \frac{\hat{E}_{k} + \hat{E}_{p} + \hat{E}_{s}}{\hat{E}_{n}} = \frac{\text{kinetic energy + potential energy + surface energy}}{\text{reference energy for normalization}} = \frac{\left[\int_{\Omega_{d}} \frac{1}{2}\rho \left|\hat{u}\right|^{2} d\hat{x}\right] + \left[\int_{\Omega_{d}} \rho g \hat{y} d\hat{x} - \frac{1}{8} \left(\rho_{g} - \rho_{l}\right) g L^{4}\right] + \left[\int_{\Omega_{d}} \sigma \left|\hat{\nabla}\phi\right| d\hat{x} - \sigma L^{2}\right]}{\left(\frac{1}{2}\rho_{l}u_{L}^{2}\right) \left(\frac{1}{2}L^{3}\right)},$$

$$(3.6)$$

where the domain of integration Ω_d is taken here to be the entire computational domain with volume $\mathcal{V}_d = L^3$. In this domain, the range of \hat{y} is defined as $\hat{y} \in [-L/2, L/2]$. The energy is computed with respect to the reference state of a quiescent air–water system with a flat interface where the water phase occupies the bottom half of the computational domain ($y \in [-1/2, 0)$) and the air phase occupies the top half ($y \in (0, 1/2]$). It is normalized by the energy \hat{E}_n of a body of water occupying half the volume of the domain and travelling uniformly at the characteristic speed $u_L = (gL)^{1/2}/(2\pi)^{1/2}$.

The total energy is observed to decay in time in general agreement with the trends observed by Wang *et al.* (2016) and Deike *et al.* (2016). Figure 3(a) suggests that the total energy is fairly insensitive to the grid spacing, so the baseline mesh should be considered sufficient for the present study. Figure 3(b) generally supports this claim as well, although some differences in the rate of change of the total energy are visible during the active wave-breaking phase due to the difference in the mesh resolution. Note, however, that the standard errors in the baseline ensemble are also noticeably larger during this phase.

Figure 3(*b*) suggests that during the large-dissipation phase $t \in (2.3, 3.1)$, the system loses energy at a rate comparable to that if the initial energy of the wave was dissipated in one wave period. The extraction of the wave period as a defining time scale is compatible with the expressions for the global Kolmogorov and Hinze scales, (2.1) and (2.2), and lends support to the dissipation rate estimate in appendix B. The rate of energy dissipation also appears to be reasonably approximated as quasi-stationary during this time interval, notwithstanding some temporal fluctuations.

The time evolution of the dissipation rate in figure 3(b) exhibits two intervals of interest. In the first interval $t \in (2.3, 3.1)$ discussed earlier, the dissipation rate remains quasi-steady as it attains its peak magnitude. In the second interval $t \in (3.5, 4.0)$, the dissipation rate gradually decays in tandem with the surrounding turbulence. The sequence of events depicted in figure 2 illustrates the importance of bubble break-up in both intervals. In § 5, the evolution of various bubble statistics in these two intervals is



Figure 3. (a) Non-dimensional ensemble-averaged total energy (E_t) of the waves in the baseline ensemble, as well as the non-dimensional total energy of one of the realizations in the ensemble with increased mesh resolution, as a function of non-dimensional time. (b) Non-dimensional rate of change of the energies in (a), computed by differencing the total energy every 500 time steps in the baseline ensemble and every 2500 time steps in the more resolved ensemble. The horizontal dotted line in (b) denotes the nominal rate of change of energy if all the energy initially present in the wave was to be dissipated in one wave period. The left-hand shaded region spans a third of a wave period between t = 2.30 and t = 3.14, while the right-hand shaded region spans a fifth of a wave period between t = 3.45 and t = 3.95. In both panels, the lines corresponding to the baseline ensemble denote ensemble-averaged quantities, while the error bars span, in each direction, twice the standard error of the energies of each realization with respect to the ensemble average, thus representing the variation over the ensemble.

analysed, and the characteristics of the bubble-mass-transfer dynamics in these intervals are compared and contrasted.

4. Algorithms

Two algorithms were used to retrieve bubble statistics from the simulation ensembles described in § 3 in order to shed light on bubble-mass transfer between large and small bubble sizes during the active wave-breaking phase. Section 4.1 describes an algorithm used to identify the sizes and locations of bubbles in each simulation snapshot. The algorithm was used to determine the bubble-size distribution and the total interfacial area. The algorithm in § 4.2 tracks these bubbles over successive simulation snapshots in order to detect break-up and coalescence events, which drive the time evolution of the bubble-size distribution as discussed in §§ 2.1 and 2.2. More details of these algorithms are discussed by Chan *et al.* (2018*a*) and Chan (2020) and a brief overview is offered in this section.

4.1. Bubble identification algorithm

An existing bubble identification algorithm (Hebert *et al.* 2008; Herrmann 2010; Tomar *et al.* 2010) was refined in this work to increase the accuracy in determining the volumes of the identified bubbles. The basis of the algorithm is the identification of connected regions of computational nodes, where each region corresponds to an individual bubble. After identifying these regions, one may then integrate the gaseous volume fraction, $1 - \phi$, over the nodes in each of these regions to obtain the total volume of each bubble. A region is assembled by linking node pairs that each satisfy a grouping criterion. The traditional criterion requires that $\phi < 1$ in each node of a pair, or that each node contains a non-zero amount of air. This criterion may create bubbles of spurious numerical origin because energetic collisions may create small wisps of air, which are sometimes also referred to as flotsam and jetsam, where $1 - \phi \ll 1$ in each of these nodes. These wisps tend to linger in the flow field due to their low buoyancy and slow degassing rate. Wisps may form

node pairs that satisfy the criterion and become spuriously linked together. Clipping small values of $1 - \phi$ mitigates this issue at the expense of the volume accuracy of resolved bubbles. Instead, this work uses the additional criterion that at least one node in a pair satisfies $1 - \phi > \phi_{c,m}$, so that nodes with small $1 - \phi$ are linked only if they neighbour nodes with large $1 - \phi$. In this work, $\phi_{c,m} = 0.5$ was selected.

The algorithm yields a list of bubbles in every simulation snapshot, and can simultaneously yield the volume and centre-of-mass (centroid) location of each identified bubble, among other statistics. Accuracy of the determined volume and centroid location is crucial to the performance of the bubble tracking algorithm to be discussed in § 4.2. Volume accuracy also impacts the trends in the bubble-size distribution to be discussed in § 5.1. A comparison of the errors incurred from various grouping criteria has been detailed in other works, such as Chan *et al.* (2018*a*) and Chan (2020). This involved systematically quantifying the incurred error for a number of simple test cases, as summarized in appendix C.

4.2. An event detection algorithm to track bubbles

A bubble tracking algorithm was developed in this work to detect break-up and coalescence events by maintaining a tally of all bubbles over time and tracing the lineage of each bubble. To construct these lineages, lists of bubbles with their sizes and locations are compared between successive simulation snapshots. These snapshots need not arise from consecutive time steps. This is because the principle of mass conservation and the Courant–Friedrichs–Lewy (CFL) condition are used to determine if a bubble has simply been advected between the two snapshots without any change in topology, or if a break-up or coalescence event has occurred. These two constraints are discussed in appendix D in relation to the errors in the identification algorithm discussed in appendix C. As discussed in § 2.1, disconnections and reconnections with the atmosphere, which is treated as a large gaseous reservoir, are not included in tallies of break-up and coalescence events, respectively. This is to prevent frequent topology changes involving the atmosphere and large, non-spherical bubbles with convoluted shapes near the wave surface from obscuring the remaining break-up and coalescence estatistics.

The time interval between successive snapshots is now discussed in relation to other characteristic time scales for the baseline ensemble introduced in § 3. The non-dimensional snapshot interval for a set of snapshots available for all 30 realizations is $\Delta t_{s,30} = 3.0 \times 10^{-2}$. A set of more closely spaced snapshots is available for 20 of these realizations with a time separation of $\Delta t_{s,20} = 6.0 \times 10^{-3}$. From (2.2) and the inertial subrange scaling $u_{L_n} \sim (\bar{\epsilon}L_n)^{1/3}$, one may estimate the characteristic mean time interval between break-up events for Hinze-scale bubbles to be $\Delta t_H \sim 10^{-1}$. This scaling for u_{L_n} also suggests that the corresponding time interval for super-Hinze-scale bubbles will exceed Δt_H . Since the snapshot interval is shorter than the mean super-Hinze-scale break-up time, the resulting statistics should provide a realistic picture of at least super-Hinze-scale turbulent break-up. In particular, there are at least O(10) snapshots in the 20-realization dataset between two super-Hinze-scale break-up events on average.

Two additional remarks are in order here. First, a snapshot interval that exceeds the simulation time step prevents the algorithm from identifying every break-up event of interest. Second, the discussion in appendix C suggests that the identification algorithm in § 4.1 cannot effectively discern the separation between two bubbles spaced less than a grid cell apart. Bubbles that break up but do not separate quickly enough may then register repeated break-up and coalescence events. An excessively small snapshot interval,

such as one that is much smaller than the integral time scale, may capture some of these confounding events. This limitation has also been observed in other event detection algorithms (Rubel & Owkes 2019) and appears to be inherent in event detection in turbulent flows. An ideal snapshot interval should resolve the characteristic mean break-up times of most bubbles while being sufficiently long to skip over these confounding events. The impacts of these algorithmic limitations are discussed in § 5.2.

5. Bubble statistics

5.1. The bubble-size distribution \overline{f}

Figure 4 plots the bubble-size distribution \overline{f} at selected time instances for the ensembles introduced in § 3. The bubble size, $D_i = [(6\mathcal{V}_i)/\pi]^{1/3}$, is the diameter of a sphere with the same volume, \mathcal{V}_i . The size distribution has dimensions $(\text{length})^{-4}$, is non-dimensionalized by the wavelength (L^4) , and was computed by histogram binning with N_{bin} bins of equal logarithmic spacing, where the smallest bin is two orders of magnitude larger than the diameter error in appendix C. The non-dimensional discrete distribution satisfies

$$\langle N_b(t)\rangle = \sum_{j=1}^{N_{bin}} \bar{f}(D_j/L; t) \Delta(D_j/L), \qquad (5.1)$$

where $\overline{f}(D_j/L; t)$ includes bubbles of sizes between D_j and D_{j+1} , and $\Delta(D_j/L) = (D_{j+1} - D_j)/L$. The distributions from the two ensembles reasonably overlap at intermediate bubble sizes, suggesting that the distribution is fairly mesh-insensitive at these sizes. While the large-bubble distribution of the more resolved ensemble has significant standard errors due to the small ensemble size and the small number of large bubbles, these large bubbles are well resolved in both ensembles.

One may link the evolution of the size distribution in figure 4 to the wave-breaking process in figure 2. As the cylindrical air cavities deform and rupture between t = 2 and t = 3 (figure 2*a*–*d*), the size distribution displays a power-law trend (figure 4*a*,*b*) with an exponent near the -10/3 value postulated by Garrett *et al.* (2000), and observed by Deane & Stokes (2002) and Deike *et al.* (2016), suggesting that the turbulent break-up cascade examined in Part 1 dominates the bubble-mass transfer during these early wave-breaking stages. As evidenced in figure 3(*b*), the dissipation rate is also large. In addition, the smallest resolvable bubbles rapidly appear, as also observed by Deane & Stokes (2002) and revisited in § 5.3. While bubbles continue to deform, fragment and interact between t = 3 and t = 4 (figure 2*c*–*f*), the power-law scaling becomes shallower at intermediate sizes (figure 4*c*,*d*). As evidenced in figure 3(*b*), the dissipation rate also begins to decay. Figure 2(*e*–*h*) suggests that large-scale air entrainment has mostly ceased by this time, even as intermediate-scale bubble break-up continues, as revisited in § 5.4.

This work focuses on the statistics of intermediate-sized (super-Hinze-scale) bubbles with non-dimensional diameters larger than 1.1×10^{-2} , or dimensional diameters larger than 2.9 mm, for which mesh insensitivity was observed. Figure 5 quantifies the evolution of the exponent of a power-law fit to the size distribution in figure 4 over an intermediate size subrange. Observe its oscillatory variation around -10/3 at early times, which may be attributed to regularity in the successive wave-surface splash-ups and impingements. This variation suggests that the largest scales may be influencing these intermediate-size statistics through analogous fluctuations in the energy and bubble-mass cascade rates. Indeed, fluctuations manifest in the dissipation rate in figure 3(b) and the bubble-mass-transfer rate, $\overline{W_b}$, in § 5.4. This variation may also reflect the finite



Figure 4. The non-dimensional bubble-size distribution $\overline{f}(D_j/L; t)$ against non-dimensional size D/L for the baseline and more resolved ensembles at (a) t = 2.50, (b) t = 2.74, (c) t = 3.71 and (d) t = 3.95. The dotted vertical lines in the same colour/shade as the respective distributions indicate the mesh resolution of the respective ensembles. The dashed sloped lines indicate the $D^{-10/3}$ power-law scaling for an idealized turbulent bubbly flow. The error bars span, in each direction, twice the standard error of the distribution in each realization with respect to the ensemble average. The shaded regions represent the bubble-size subrange over which the power-law fit exponents in figures 5 and 6 were obtained. More snapshots in time of the bubble-size distribution are provided by Chan (2020).

convergence time of the break-up dynamics towards quasi-equilibrium (Filippov 1961; Qi, Masuk & Ni 2020). Figure 6(*a*) depicts the size distribution averaged over these early time instances with two power-law fits, which are in general agreement with $D^{-10/3}$, although the fit exponent exhibits slight sensitivity to the size subrange used for fitting. A similar agreement was obtained in the ensemble-averaged and time-averaged statistics of Deane & Stokes (2002) and Deike *et al.* (2016), and the time-averaged statistics of Wang *et al.* (2016), which builds confidence in the results of this work. Ensemble averaging improves statistical convergence and reduces data scatter such as that observed by Wang *et al.* (2016) in their instantaneous distributions. While the instantaneous distribution in this work oscillates between D^{-3} and D^{-4} , better agreement with $D^{-10/3}$ is obtained via time averaging over the entire interval, consistent with the suggestion of Deike *et al.* (2016) that the scaling of the time-averaged distribution, f^T , is sensitive to the time-averaging window, as is revisited in § 5.4. A more rigorous test of the $D^{-10/3}$ scaling is shown in figure 7(*a*) by premultiplying the time-averaged distribution and using a linear scale on the vertical axis. Time averaging is further explored in appendix A.

Returning to figure 5, the fit exponent deviates from -10/3 at later times, about one wave period after the onset of breaking. Interestingly, the exponent does not oscillate, in seeming correlation with the cessation of large-scale entrainment. Figure 6(*b*) depicts the size distribution averaged over these late time instances. Its increased magnitude at most



Figure 5. The variation of the exponent of a power-law fit over a subset of the baseline distribution, \bar{f} , in figure 4 with non-dimensional time. The fit was performed over bubbles with non-dimensional diameters between 1.07×10^{-2} and 2.23×10^{-2} . This bubble-size subrange is marked in the shaded regions in figure 4. The error bars denote twice the standard error in the fit exponent over the baseline ensemble. The shaded regions span the same time intervals as the corresponding regions in figure 3(b).



Figure 6. The non-dimensional time-averaged size distribution $\overline{f}^T(D_j/L)$ against non-dimensional size D/L for both ensembles. The time-averaging interval is (a) between t = 2.30 and t = 3.14, corresponding to the left-hand shaded regions in figures 3(b) and 5, and (b) between t = 3.45 and t = 3.95, corresponding to the right-hand shaded regions in the same figures. For a description of the vertical lines, dashed sloped line in (a) and error bars, refer to the caption of figure 4. The bubble-size subrange over which the bottom power-law fit in (a) and the power-law fit in (b) were performed is the same subrange highlighted in figure 4 and used in figure 5. The subrange for the top power-law fit in (a) is $[1.69 \times 10^{-2}, 3.54 \times 10^{-2})$. The fit exponent uncertainties denote twice the standard error over the baseline ensemble.

resolved sizes relative to figure 6(a) (by about 3–5 times; see also figure 15) suggests that there are more small and intermediate-sized bubbles at later times, and the break-up flux from larger sizes consistently outweighs mass loss from degassing. Two distinct power-law scalings emerge in two size subranges in agreement with the measurements of Tavakolinejad (2010) and Masnadi *et al.* (2020), which builds further confidence in the results of this work (see also Castro, Li & Carrica 2016). The size distribution is reasonably described by a $D^{-8/3}$ scaling at intermediate sizes, which is shallower than $D^{-10/3}$. This is supported by the compensated time-averaged distribution in figure 7(*b*). At large sizes, the power-law scaling is steeper than $D^{-8/3}$ by a factor of D^{-2} in agreement with the increased importance of buoyant degassing postulated by Garrett *et al.* (2000) and observed by Deike *et al.* (2016), as well as in figure 2(*e*–*h*). For comparison, Tavakolinejad (2010) obtained



Figure 7. The compensated time-averaged size distribution using the same time-averaging intervals as in figure 6: (a) $\bar{f}^T(D_j/L)D_j^{10/3}$ from the first interval, premultiplied by the inverse of the $D^{-10/3}$ scaling, and $(b) \bar{f}^T(D_j/L)D_j^{8/3}$ from the second interval, premultiplied by the inverse of the observed $D^{-8/3}$ scaling, against non-dimensional size D/L for both simulation ensembles. The compensated distribution is normalized by its value at $D/L = 1.07 \times 10^{-2}$. For a description of the vertical lines and error bars, refer to the caption of figure 4.

an exponent between -2.85 and -2.58 at small sizes, and between -6.57 and -3.85 at large sizes.

Figures 3(b) and 4-7 reflect the unsteadiness in the wave dynamics and demonstrate the emergence of two time intervals with distinct characteristics, suggesting the presence of distinct bubble generation and evolution mechanisms. In the first interval, the dissipation rate appears to be quasi-steady, and the $D^{-10/3}$ power-law scaling in the time-averaged size distribution supports the presence of a quasi-steady bubble break-up cascade. In the second interval, the dissipation rate is seen to decay, and the time-averaged size distribution deviates from $D^{-10/3}$. Yet, the robustness of an alternative power-law scaling with a negative exponent is indicative of size-invariant cascade-like behaviour, as Part 1 suggested that size-local break-up may still occur in the absence of the $D^{-10/3}$ scaling. These observations motivate a closer look at other bubble statistics during these two intervals to test the theoretical analysis developed in Part 1 for the break-up cascade in the first interval, and to extend this analysis to the possible cascade-like behaviour in the second interval. Note that these conclusions were drawn from statistics over a limited size range. Future ensembles with larger scale separation will be valuable for shedding more light on these conclusions. Volume averaging also precludes the reporting of spatial size distributions. The physical parameters and grid size selected in this work reflect the trade-off between the resolved size range and the ensemble size. As noted in \S 1 and 3.2, a larger ensemble increases statistical convergence in a statistically unsteady flow that does not strictly permit time averaging. However, it also necessitates a smaller resolved range from a practical perspective. The mesh insensitivity in figures 4, 6 and 7, as well as the agreement of their power-law scalings with prior works, supports these choices in light of this trade-off.

5.2. The differential break-up rate $\overline{g_b f}$

In order to gain more insights into the evolution of the size distribution, \bar{f} , as described by (2.4) and (2.6), the differential break-up rate $\overline{g_{b}f}$ is now analysed. Break-up events were detected in the baseline ensemble using the algorithm in § 4.2. The ensemble-averaged event rate per unit domain volume and unit size is averaged over the non-dimensional



Figure 8. The normalized differential break-up rate $\widetilde{g_b f}(D_j/L; t)$ averaged over the non-dimensional sampling interval $16\Delta t_{s,20} = 9.6 \times 10^{-2}$ plotted against non-dimensional size D/L for a 20-realization baseline ensemble subset at (a) t = 2.51, (b) t = 2.75, (c) t = 3.71 and (d) t = 3.95. The rate is normalized such that the maximum value in each plot is 1, i.e. $\widetilde{g_b f} = \overline{g_b f}/\overline{g_b f}|_{max}$. The selected bin sizes are twice those in figure 4 in logarithmic space. For a description of the vertical lines and error bars, refer to the caption of figure 4. The dashed sloped lines indicate the theoretical D^{-4} power-law scaling for an idealized turbulent bubbly flow. The shaded regions represent the bubble-size subrange over which the power-law fit exponents in figures 9 and 10 were obtained. More snapshots in time are provided by Chan (2020).

sampling interval $16\Delta t_{s,20} = 9.6 \times 10^{-2}$ to converge statistics over more events. This interval is kept small enough that temporal variations are not significantly smoothed. The number of histogram bins for $\overline{g_{bf}}$ is half that for \overline{f} to increase the bin width and improve this convergence. To ensure that the snapshot interval discussed in § 4.2 has been reasonably selected, rates obtained using two intervals, $\Delta t_{s,20}$ and $2\Delta t_{s,20}$, are reported to demonstrate snapshot convergence. Since the variation of $\overline{g_{bf}}$ with parent bubble size is of interest, $\overline{g_{bf}}$ is normalized by its maximum value $(\overline{g_{bf}})$ at every time instance to highlight this convergence. This normalization is also carried out in § 5.4.

Figure 8 shows $\overline{g_b f}$ near the time instances for which \overline{f} was plotted in figure 4. Signatures of the D^{-4} scaling derived for a quasi-steady turbulent break-up cascade in Part 1 emerge at intermediate sizes. Note that the computation of $\overline{g_b f}$ samples energetic regions more frequently since break-up only occurs with sufficient energy to change the bubble topology. Thus, $\overline{g_b f}$ is susceptible to large-scale inhomogeneities, and is difficult to converge since the scale separation in the simulated waves is not exceedingly large.

Figure 9 quantifies the evolution of the exponent of a power-law fit on the differential rate in figure 8 over a subrange of intermediate sizes, relative to the -4 exponent derived in Part 1 using traditional turbulent-flow scalings. The behaviour of the exponent in figure 9 for \bar{g}_{bf} loosely tracks the corresponding behaviour in figure 5 for \bar{f} : in the early



Figure 9. The variation of the exponent of a power-law fit over a subset of the differential break-up rate, $\overline{g_b f}$, in figure 8 with non-dimensional time. The fit was performed on the data with a snapshot interval of $\Delta t_{s,20}$. For a description of the bubble-size subrange over which the fit was performed, error bars and shaded regions, refer to the caption of figure 5.



Figure 10. The normalized time-averaged differential break-up rate $\widetilde{g_{bf}}^T(D_j/L)$ against non-dimensional size D/L for the 20-realization baseline ensemble subset. Panels (a,b) correspond to the time-averaging intervals in figure 6. For a description of the vertical and sloped lines, error bars, normalization, bubble-size subrange and dataset over which the fits were performed, and fit-exponent uncertainties, refer to the captions of figures 4, 6, 8 and 9.

wave-breaking stages, the exponent oscillates around the exponent discussed in Part 1; in the late stages, it does not strongly oscillate. As with the size distribution in figure 6, the differential rate may be time-averaged over these two intervals, as depicted in figure 10. In the first interval, it may be described by a $D^{-4.1\pm0.3}$ power-law fit, in agreement with D^{-4} . In the second interval, it may be described by a $D^{-3.3\pm0.4}$ fit, which is about a factor of $D^{-2/3}$ steeper than the $D^{-8/3}$ scaling in the corresponding size distribution, although vestiges of the D^{-4} scaling remain in the differential rate at larger sizes. These power-law scalings at intermediate sizes are largely consistent with the turbulent scaling of the break-up frequency, $g_b \sim D^{-2/3}$, in agreement with the references cited in § 1 that were used in the theoretical analysis of Part 1.

In summary, the trends observed in $\overline{g_b f}$ mostly echo those for \overline{f} in § 5.1. The unusual persistence of the D^{-4} scaling may be suggestive of a tendency of the bubble-mass flux to remain quasi-local and quasi-self-similar, since $\overline{W_b} \sim \overline{g_b f} D^4$ in this limit as shown in Part 1. These characteristics of the bubble-mass flux are further addressed in § 5.4.

Analogous to § 5.1, future ensembles of higher Re_L and We_L simulations could bring about more clarity on the robustness of these scalings.

5.3. The break-up probability distribution \check{q}_b

In addition to $\overline{g_{bf}}$, the probability distribution of child bubble sizes, \check{q}_b , also drives the evolution of \bar{f} , as described by (2.4) and (2.6), and may also be computed by the algorithm in § 4.2. The distribution $\check{q}_b(D_c^3; t|D_p^3)$ is symmetric in bubble-volume space and is presented in this space for a more intuitive interpretation. It satisfies

$$2 = \sum_{j=1}^{N_{bin}} \check{q}_b \left(D_{c,j}^3; t | D_p^3 \right) \Delta \left(D_{c,j}^3 / D_p^3 \right).$$
(5.2)

The data for each parent bubble size $D_{p,j}$ are averaged over two histogram bins $[D_{p,j}, D_{p,j+2})$, where the bins are identical to those in § 5.2. The reported \check{q}_b is time-averaged over the two time intervals identified in §§ 3.3, 5.1 and 5.2.

Figure 11 plots the time-averaged break-up probability distribution, $\overline{\check{q}}_b^T$, for three distinct parent bubble sizes and the two aforementioned time intervals. Each row corresponds to a single parent bubble size, while each column corresponds to a single time interval. Some child bubble sizes are smaller than the mesh resolution and have to be excluded. Under the influence of this exclusion, $\overline{\check{q}}_b^T$ falls off towards large and small child bubble sizes for small parent bubbles.

In general, the child bubble volume distribution is relatively close to uniform for events within the intermediate bubble-size subrange where power-law fits were earlier obtained for \overline{f} and $\overline{g_{bf}}$ in §§ 5.1 and 5.2. Hence, the uniform distribution, which was analysed in \overline{f}

Part 1, appears to be a suitable, albeit rudimentary, surrogate model for $\overline{\check{q}_b}^T$ in most cases. One exception occurs at large parent bubble sizes outside the aforementioned size subrange in the early wave-breaking stages. Here, large-size-ratio events involving one large child bubble and one small child bubble do frequently occur as evidenced in figure 11(e), bypassing the break-up cascade and generating small bubbles as observed in § 5.1. These individual non-local contributions to bubble-mass transfer are small in volume and may not necessarily influence the locality of the break-up flux, $\overline{W_b}$, as noted in appendix B of Part 1. Their relative influence is analysed in more detail in § 5.4.1. Another exception occurs in the late wave-breaking stages in figure 11(d). In these two cases, the break-up probability is reasonably described by a beta-distribution fit, which was also discussed in Part 1 as a potential surrogate model. These exceptions indicate that it does not seem appropriate to characterize a turbulent bubbly flow with a single distribution shape without giving more consideration to the bubble-size subrange or flow stage of interest, as many previous studies have done. For example, experimental studies typically report a single distribution of child bubble volumes, agglomerating data from different parent bubble sizes (see figure 1 of Qi et al. (2020), and references therein). This has contributed to a multitude of existing models for the child bubble volume distribution, as noted in Part 1. Interestingly, the results of this work are in qualitative agreement with the model of Lehr, Millies & Mewes (2002), where 'equal sized breakage is preferred for small bubbles ... as the size of the parent bubbles increases, unequal breakage is preferred'.

5.4. The break-up flux $\overline{W_b}$

The results of § 5.1 revealed that the bubble-size distribution dramatically evolves between t = 2 and t = 4. Indirect evidence of a quasi-steady bubble break-up cascade (Garrett *et al.*



Figure 11. The time-averaged break-up probability $\overline{\dot{q}}_{b}^{-T}(D_{c,j}^{3}|D_{p}^{3})$ against normalized child bubble volume D_{c}^{3}/D_{p}^{3} for the 20-realization baseline ensemble subset during the (a,c,e) early (t = 2.30-3.14) and (b,d,f) late (t = 3.45-3.95) wave-breaking stages. The non-dimensional parent bubble sizes considered are (a,b) between 1.86×10^{-2} ($We_{D_{p}} = 18$) and 2.23×10^{-2} ($We_{D_{p}} = 25$), (c,d) between 2.68×10^{-2} ($We_{D_{p}} = 33$) and 3.23×10^{-2} ($We_{D_{p}} = 45$) and (e,f) between 3.54×10^{-2} ($We_{D_{p}} = 53$) and 4.25×10^{-2} ($We_{D_{p}} = 72$). For the definition of $We_{D_{p}}$, refer to (2.4) in Part 1. Only child bubbles of radii larger than the mesh resolution are considered. The vertical dashed lines demarcate the child bubble volumes in the break-up event where one of the child bubble radii corresponds to this resolution limit. The error bars denote one standard error. The time intervals in the legend denote the snapshot intervals in the detection algorithm. Fits to the beta distribution have been added in (d,e). These fits were performed on the data obtained by taking the snapshot interval to be $\Delta t_{s,20}$, which was also used to plot the histogram bars. The two numbers in the corresponding legend entries are the shape parameters characterizing each fit, which were obtained via the method of moments.

2000) was observed in the early wave-breaking stages, although multiple mechanisms could explain the observed $D^{-10/3}$ power-law scaling in the size distribution (e.g. Yu *et al.* 2019; Yu, Hendrickson & Yue 2020). The present simulations, with the accompanying detection algorithm for break-up events, also provide detailed break-up statistics, which are analysed in §§ 5.2 and 5.3. These statistics contribute to the break-up flux, $\overline{W_b}(D; t)$, as discussed in Part 1 and summarized in § 2.1, and provide further indirect evidence

of a break-up cascade. The break-up flux across size D can be directly decomposed into incoming contributions $\overline{I_p}(D_p; t|D)$ from parent bubbles of sizes $D_p > D$ and outgoing contributions $\overline{I_c}(D_c; t|D)$ to child bubbles of sizes $D_c < D$, as shown in § 2.2, as well as figures 4 and 5 of Part 1. These complementary decompositions respectively provide information on infrared and ultraviolet locality in the flux. Satisfying these measures of locality is a necessary condition for the presence of a break-up cascade, with $\overline{I_p}$ and $\overline{I_c}$ decaying rapidly away from D. Cascade-like behaviour may also be directly characterized by size invariance and quasi-stationarity of the break-up flux at intermediate sizes. These observations are direct consequences of the theoretical analysis performed in Part 1. In this subsection, the simulation results are further leveraged by examining these quantities to obtain more direct observations of cascade-like behaviour.

5.4.1. Infrared and ultraviolet locality

Figure 12 plots the time-averaged differential incoming (parent, $\overline{I_p}^T$) and outgoing (child, $\overline{I_c}^T$) contributions to the break-up flux for the two time intervals identified in §§ 3.3, 5.1 and 5.2, and an intermediate cutoff bubble size near $D/L = 2 \times 10^{-2}$. This choice of D/L is representative of a bubble size within the intermediate-size subrange in figures 4–10 in which the break-up cascade scalings were recovered during the first time interval of interest. Other choices of D/L, which are not shown here, display a similar behaviour. The differential incoming contributions from parent bubbles strongly decay with increasing parent bubble size, while the differential outgoing contributions to child bubbles strongly decay with decreasing child bubble size, indicating that the break-up flux is strongly infrared and ultraviolet local, as suggested in Part 1. Power-law fits of these decay rates exhibit exponents near -7 and 5 for $\overline{I_p}^T$ and $\overline{I_c}^T$, respectively. The theory in Part 1 predicts these exponents for a uniform child bubble volume distribution, q_b , that remains invariant with parent bubble size. Slight deviations from these idealized values may be accounted for by the mild deviations of $\overline{\check{q}_b}^T$ from uniformity and parent-bubble-size invariance observed in § 5.3. Note that reasonable agreement with the exponents from Part 1 is obtained for both time intervals even though the $D^{-10/3}$ scaling is absent from the size distribution in the second interval. This suggests that cascade-like behaviour persists in this interval, even as the dissipation rate decays. The recovery of the exponents in the second interval is consistent with $\overline{g_{bf}} \sim D^{-4}$, whose persistence in both intervals was noted at the end of § 5.2.

One might observe that the decay rate of $\overline{I_p}^T$ becomes gentler at very large parent bubble sizes, especially during the first time interval. This is reminiscent of the occurrence of large-size-ratio cascade-bypassing break-up events in large parent bubbles in figure 11(*e*). Two remarks are in order here. First, the contributions from these events to the break-up flux are orders of magnitude smaller than the transfers from events of highest locality. Second, these contributions arise from intermittent break-up events as evidenced in figure 13, which illustrates the instantaneous differential incoming contributions from parent bubbles at two time instances during the first interval. Notwithstanding these caveats, these results suggest that the break-up dynamics is well approximated as local in bubble-size space in both time intervals of interest.

5.4.2. Size variation and time evolution of the break-up flux

Figure 14 plots the variation of the time-averaged break-up flux $\overline{W_b}^T(D)$ with cutoff bubble size *D* for the two time intervals identified in §§ 3.3, 5.1 and 5.2. Observe that



Figure 12. (a,c) The normalized time-averaged differential outgoing transfer rate $\tilde{L}_c^T(D_{c,j}|D)$ to child bubbles as a function of non-dimensional child bubble size D_c/L and (b,d) the corresponding differential incoming transfer rate $\tilde{L}_p^T(D_{p,j}|D)$ from parent bubbles as a function of non-dimensional parent bubble size D_p/L , for the 20-realization baseline ensemble subset at the non-dimensional cutoff size $D/L = 1.86 \times 10^{-2}$ marked by the dashed vertical line. These statistics are time-averaged over the (a,b) early (t = 2.30-3.14) and (c,d) late (t = 3.45-3.95) wave-breaking stages. The normalization, histogram bins, shaded regions and dotted vertical lines are identical to those in figure 8. Break-up events involving subgrid child bubbles are excluded. The error bars and fit-exponent uncertainties denote twice the standard error over the ensemble. The power-law fits were performed over bubbles with non-dimensional diameters between 9.74×10^{-3} and 1.69×10^{-2} for \tilde{L}_c^T , and

between 2.04 × 10⁻² and 3.54 × 10⁻² for \tilde{I}_p^T , on the dataset with snapshot interval $\Delta t_{s,20}$.



Figure 13. The normalized differential incoming transfer rate $\widetilde{I_p}(D_{p,j}; t|D)$ from parent bubbles as a function of non-dimensional parent bubble size D_p/L for the 20-realization baseline ensemble subset at the non-dimensional cutoff size $D/L = 1.86 \times 10^{-2}$, and at the times (a) t = 2.51 and (b) t = 2.99. These statistics are averaged over the non-dimensional sampling interval $16\Delta t_{s,20} = 9.6 \times 10^{-2}$ used in figure 8. For a description of the normalization, shaded regions, vertical lines, excluded events, error bars, bubble-size subrange and dataset for the power-law fits, and the fit-exponent uncertainties, refer to the caption of figure 12.



Figure 14. The normalized time-averaged break-up flux $\widetilde{W_b}^T(D/L)$ against non-dimensional size D/L for the 20-realization baseline ensemble subset. These statistics are time-averaged over the (*a*) early (t = 2.30-3.14) and (*b*) late (t = 3.45-3.95) wave-breaking stages. For a description of the normalization, vertical lines, excluded events and error bars, refer to the caption of figure 12. The shaded regions depict the bubble-size subrange over which the size-averaged break-up flux in figure 15 was obtained.

there is some degree of size invariance in the break-up flux in both time intervals, even though the scale separation in the simulated waves is not exceedingly large as remarked in §§ 5.1 and 5.2. This size invariance is expected in the theory from Part 1 if $I_p \sim D_p^{-7}$ and $I_c \sim D_c^5$, or more generally if the magnitude of the power-law exponent for I_p is larger than that for I_c by 2. These exponents are approximately recovered for both time intervals in § 5.4.1. Note that the size invariance of the break-up flux, $\overline{W_b}$, is also consistent with approximate stationarity in the size distribution, \bar{f} , as implied by (2.5), as well as the usage of the break-up flux as a proxy for the entrainment flux, as discussed in $\S 2.1$. Further implications of the consistency of quasi-self-similarity and quasi-stationarity in the break-up dynamics are explored in appendix A with reference to the discussion on time averaging in § 5.1. Finally, observe that the ratio of the flux at intermediate sizes to the flux at large sizes differs in the two time intervals of interest. The ratio is close to unity in the first interval, with variations due to large-scale inhomogeneities, but increases by an order of magnitude in the second interval. This suggests that active entrainment at the large scales directly drives bubble-mass transfer from large- to small-bubble sizes in the first interval. The entrainment appears to have ceased in the second interval as mentioned in \S 5.1, and bubble-mass transfer seems to be chiefly the result of the fragmentation of already-entrained large cavities and bubbles.

Figure 15 plots the variation of the flux $\overline{W_b}^D(t)$ with time, averaged over a narrow intermediate range of bubble sizes enveloping the cutoff size considered in § 5.4.1. Because of the quasi-size-invariance observed in figure 14, the results of figure 15 are representative of the intermediate-size dynamics considered in this work. The flux has distinct characteristics in the two time intervals identified in §§ 3.3, 5.1 and 5.2. The first time interval with a relatively constant rate of energy dissipation appears to be concomitant with an approximately constant flux, closing the loop on the presence of a quasi-steady turbulent bubble break-up cascade in this interval. The oscillations in the flux mirror the oscillations in the dissipation rate in figure 3(b), and appear to be related to the earlier discussion in § 5.1 on the oscillating power-law exponent for the size distribution. The second time interval with a decreasing dissipation rate appears to be associated with a flux that linearly increases with time. As discussed in § 5.4.1, the flux was observed to be strongly size local during this interval, which is indicative of the persistence of



Figure 15. The normalized size-averaged break-up flux $\widetilde{W}_b^D(t)$ against non-dimensional time *t* for the 20-realization baseline ensemble subset. The flux is averaged over the bubble-size subrange spanning non-dimensional diameters between 1.69×10^{-2} and 2.23×10^{-2} , as indicated by the shaded regions in figure 14. For a description of the normalization, excluded events and error bars, refer to the caption of figure 12. The shaded regions span the same time intervals as those in figures 3(*b*), 5 and 9.

cascade-like behaviour. However, the temporally increasing flux does not drive a uniform increase in the magnitude of the size distribution at all sizes, as depicted in figures 4 and 6. Instead, it is observed in § 5.1 that the power-law scaling of the size distribution transitions from $D^{-10/3}$ to $D^{-8/3}$. This suggests that small bubbles may be depleted more rapidly at these late times. The continuation of the flux over several characteristic times suggests that the volume of gas present in small bubble sizes gradually increases over time. The assumption in Part 1 that one may neglect this accumulation should eventually break down towards the late wave-breaking stages. The increased importance of coalescence due to this accumulation, and its potential role in the aforementioned depletion of small bubbles, is addressed in appendix E.

6. Conclusions

This paper investigates the evolution of the bubble-size distribution and other bubble statistics beneath a breaking wave during the active wave-breaking phase using ensembles of numerical simulations. A large ensemble of simulations enhances statistical convergence in these bubble statistics, which is particularly important for the time-dependent behaviour explored in this paper. Two time intervals of interest are identified, with comparison and contrast between them throughout the paper. This analysis goes beyond the size distribution by identifying and characterizing individual break-up events in order to directly observe characteristics associated with the bubble-mass cascade phenomenology that were identified in Part 1. The evolving dynamics in the bubble-mass-transfer process is summarized in table 2. The results of this paper (Part 2) suggest that cascade-like behaviour is present in both time intervals, even when the equilibrium $D^{-10/3}$ power law is not observed in the size distribution at intermediate sizes. They highlight that different bubble generation and evolution mechanisms emerge at different length and time scales during the wave-breaking process, even within the relatively limited range of length scales captured in these simulations.

Stage	Nature of flux $\overline{W_b}$	Nature of dissipation $\bar{\varepsilon}$	$\begin{array}{c} \text{Scaling} \\ \text{of} \bar{f} \end{array}$	Locality of flux $\overline{W_b}$	Description of bubble-mass transfer
Early Late	Quasi-steady Increasing	Quasi-steady Decreasing	$D^{-10/3}$ $D^{-8/3}$	Strong Strong	Bubble break-up cascade Cascade-like behaviour + other dynamics

Table 2. Summary of bubble-mass-transfer dynamics in the active wave-breaking phase.

of higher Re_L and We_L simulations with resolution of a larger range of scales may bring about more clarity on these mechanisms.

The evolution of the bubble-size distribution and other associated bubble statistics in the present simulation ensembles may be summarized as follows. In the early wave-breaking stages, large air cavities are entrained and successively fragmented. The corresponding rate of bubble-mass transfer from large- to small-bubble sizes is quasi-steady, as is the energy dissipation rate, which also reaches a maximum value in this time interval. These are the ideal characteristics of a forward bubble-mass cascade driven by turbulent fragmentation. As originally proposed by Garrett *et al.* (2000), this is accompanied by a $D^{-10/3}$ power-law scaling in the size distribution. The break-up statistics of Part 2 provide strong support for the theoretical results of Part 1 in this early time interval. The time-averaged size distribution matches the $D^{-10/3}$ scaling in agreement with Deane & Stokes (2002), Wang et al. (2016) and Deike et al. (2016). Deviations of the instantaneous size distribution from this scaling are observed but are reduced by time averaging over the entire interval. The D^{-4} power-law scaling derived in Part 1 for the differential break-up rate is also recovered. Notwithstanding several intermittent non-local events for large parent bubbles, the probability distribution of child bubble volumes is roughly uniform in the range of parent bubble sizes where the $D^{-10/3}$ scaling is observed in the time-averaged size distribution. The resulting break-up flux remains strongly infrared and ultraviolet local. The incoming differential flux from parent bubbles approaches a D_p^{-7} power-law scaling over a range of parent bubble sizes, while the outgoing differential flux to child bubbles approaches a D_c^5 power-law scaling over a range of child bubble sizes. These scaling exponents were predicted in Part 1 for a uniform child bubble volume distribution. The observed scalings suggest that the underlying bubble-mass-transfer process is indeed size local and self-similar, although a more robust demonstration of self-similarity may only be evident in larger waves of higher Re_L and We_L . Together, these observations provide strong support for the presence of a quasi-steady turbulent bubble break-up cascade at intermediate bubble sizes during the early wave-breaking stages.

In the late wave-breaking stages, large entrained bubbles continue to break up into smaller bubbles. The corresponding rate of bubble-mass transfer from large to small sizes accelerates while the dissipation rate begins to decay. Two distinct power-law scalings appear in the size distribution in agreement with Tavakolinejad (2010) and Masnadi *et al.* (2020) but in deviation from the $D^{-10/3}$ scaling. The increased magnitude of the size distribution suggests that small and intermediate bubbles are more populous in this time interval. Vestiges of locality, stationarity and self-similarity are nonetheless present in the corresponding break-up statistics, indicating that the break-up dynamics still exhibits cascade-like behaviour. The steepening of the size distribution at large bubble sizes is indicative of buoyant degassing. A shallower size distribution appears at intermediate

bubble sizes. The data appear to be consistent with a more rapid depletion of small bubbles, potentially via coalescence.

Several concluding remarks are made in view of these observations. The theoretical framework developed in Part 1 enables one to go above and beyond the traditional $D^{-10/3}$ power-law scaling in the size distribution in diagnosing the presence of a break-up cascade. The results of Part 2 demonstrate that the access to detailed spatial information provided by numerical simulations allows a more precise evaluation of various components of the theoretical framework. The multifaceted nature of this analysis enables a more nuanced description of the underlying break-up dynamics even when the $D^{-10/3}$ scaling is absent. For example, a number of key physical aspects of the cascade phenomenology, such as locality, were determined to be present even in the late wave-breaking stages. The elements of the formalism in Part 1 could thus be seen as an analytical toolkit that may be used to selectively probe the characteristics of break-up and coalescence dynamics in multiscale two-phase flows. This toolkit may be used to advance understanding of cascading processes, and more generally to inform the validation and development of model kernels for the population balance equation.

Knowledge of the physical mechanisms underlying bubble generation enables the quantification of interfacial fluxes and radiation scattering, which are of practical importance in maritime and climate studies, such as carbon sequestration and the persistent wake signatures of seafaring vessels. The rich temporal evolution of the bubble-mass-transfer process observed in this work reflects the complexity in modelling these mechanisms. Elucidation of sustained cascade-like behaviour in the bubble-mass-transfer dynamics presents opportunities for modelling of subgrid bubble break-up in large-eddy simulations.

Acknowledgements. With a deep sense of gratitude, we dedicate this paper and its companion (Part 1) to the memory of Professor Juan Lasheras. These manuscripts build on fundamental concepts introduced in his early work on turbulent two-phase flows. His untimely passing cut short a brilliant career, but his seminal contributions in the fields of fluid mechanics and biomedical engineering will prove a lasting legacy. The authors acknowledge computational resources from the US Department of Energy's INCITE Program. The authors are grateful to A. Mani for fruitful discussions on the bubble-size distribution and to M.S. Dodd for joint work on algorithms to retrieve statistics from numerical simulations that inspired this work.

Funding. This investigation was funded by the Office of Naval Research, Grant no. N00014-15-1-2726, and is also supported by the Advanced Simulation and Computing programme of the US Department of Energy's National Nuclear Security Administration via the PSAAP-II Center at Stanford University, Grant no. DE-NA0002373. W.H.R.C. is also funded by a National Science Scholarship from the Agency for Science, Technology and Research in Singapore.

Declaration of interests. The authors report no conflict of interest.

Author ORCIDs.

- Wai Hong Ronald Chan https://orcid.org/0000-0002-3525-6319;
- Perry L. Johnson https://orcid.org/0000-0002-7929-9396;
- Parviz Moin https://orcid.org/0000-0002-0491-7065;
- ^D Javier Urzay https://orcid.org/0000-0003-3675-8377.

Appendix A. Averaging operations for computing bubble statistics

The ensemble-averaged bubble-size distribution f(x, D; t) was defined in Part 1 at every location x, for every bubble size D and at some time t by ensemble averaging $(\langle \cdot \rangle)$ the number density function of a bubble population, which is constructed by summing a contribution from each bubble in the population $i = 1, \ldots, N_b(t)$ having a centroid

location x_i and an equivalent size D_i . Then f may be expressed as

$$f(\mathbf{x}, D; t) = \left\langle \sum_{i=1}^{N_b(t)} \delta\left(\mathbf{x} - \mathbf{x}_i(t)\right) \delta\left(D - D_i(t)\right) \right\rangle.$$
(A1)

It was alluded to in §2 that the volume-averaged and ensemble-averaged size distribution $\bar{f}(D; t)$ is typically reported in breaking-wave simulations and experiments as

$$\bar{f}(D;t) = \frac{1}{\mathcal{V}} \int_{\Omega} \mathrm{d}\mathbf{x} f\left(\mathbf{x}, D; t\right) = \frac{1}{\mathcal{V}} \left\langle \sum_{i=1}^{N_b(t)} \delta\left(D - D_i(t)\right) \right\rangle,\tag{A2}$$

where $\mathcal{V} = \int_{\Omega} d\mathbf{x}$ is the volume of the domain Ω over which the bubble population is integrated. Domain Ω should be selected such that it always contains all $N_b(t)$ bubbles. In this work, \mathcal{V} is chosen to be the characteristic wave volume, L^3 . Since this coincides with the computational domain volume, all $N_b(t)$ bubbles are thereby always included. The volume-averaging operation is analogously defined for other bubble statistics like $\overline{g_{bf}}$. The ensemble-averaging and volume-averaging operations commute when \mathcal{V} and Ω are identical across all the ensemble realizations. Thus, \overline{f} satisfies the normalization conditions

$$\frac{\langle N_b(t)\rangle}{\mathcal{V}} = \langle n_b(t)\rangle = \int_0^\infty \mathrm{d}D\bar{f}(D;t),\tag{A3}$$

where $n_b(t)$ is the number of bubbles in the system per unit domain volume for an ensemble realization, or the global bubble number density in the realization. The population balance equation for $f(x, D; t)D^3$ was written in Part 1 as

$$\frac{\partial [f(\mathbf{x}, D; t)D^3]}{\partial t} + \frac{\partial [v_i(\mathbf{x}, D; t)f(\mathbf{x}, D; t)D^3]}{\partial x_i} + \frac{\partial [v_D(\mathbf{x}, D; t)f(\mathbf{x}, D; t)D^3]}{\partial D}$$

$$= H(\mathbf{x}, D; t), \qquad (A4)$$

where v_i and v_D are the velocities of fD^3 in the *x*–*D* phase space along the spatial and bubble-size dimensions, respectively, and *H* is a model term that includes source, sink and non-local transfer terms for the transport of fD^3 . Volume averaging the equation over a domain that always includes all $N_b(t)$ bubbles yields (2.3). A similar procedure may be used to obtain (2.4). Note that the spatial and bubble-size dimensions are orthogonal in the *x*–*D* phase space. Thus, the expected *D*-scaling of a quantity, such as the theoretical variations of f(x, D; t) and $\bar{f}(D; t)$ with *D*, should remain invariant under volume averaging provided there is sufficient statistical convergence in the quantity.

Volume averaging aggregates statistics at different locations, and provides a convenient way of achieving statistical convergence in a limited number of realizations, as well as a more concise description of the underlying dynamics (Hinze 1955). It is argued here that volume averaging also achieves a good approximation for low-order bubble statistics, including the size distribution, as if the underlying flow was statistically homogeneous (Deike *et al.* 2016). More specifically, it averages these statistics over small, localized regions that may each be treated as statistically homogeneous in the spirit of local isotropy. This may be motivated by the arguments behind Kolmogorov's refinements to his original energy cascade hypotheses (Kolmogorov 1962). The original hypotheses (Kolmogorov 1941) are best applied in regions where the assumption of spatial homogeneity is robust.

In a statistically inhomogeneous flow, diffusion fluxes across large-eddy length and time scales may become important, thereby manifesting as large-scale variations in the turbulent kinetic energy, as well as intermittency in the localized production and dissipation rates. The refinements of Kolmogorov (1962) attempt to characterize these variations systematically through an assumption on their statistical distribution. To enable this, volume averaging is performed in small, localized regions that are each approximately statistically homogeneous. The original hypotheses are then recovered for locally volume-averaged flow statistics, with constants of proportionality that account for large-scale inhomogeneities. The original and refined hypotheses do not yield significantly different results when applied to low-order flow statistics (Pope 2000, § 6.7.4), such as the second-order velocity structure functions. The global volume-averaging procedure adopted in this work eliminates these diffusion fluxes by averaging flow and bubble statistics over a volume where no net transport in or out is expected. This assumes each small, localized region in the global averaging volume is subject to the same characteristic large-eddy length and time scales, in the spirit of the original hypotheses of Kolmogorov (1941). This is thus likely to be reasonable for low-order flow and bubble statistics, with additional corrections necessary for higher-order statistics when, for example, $\bar{\varepsilon}^n$ is no longer well approximated by $(\bar{\varepsilon})^n$ for large n. Hence, volume averaging is a physically relevant way of computing low-order bubble statistics in addition to its expedience. Nevertheless, knowledge of the spatial distribution of bubbles may at times be important, and two representative snapshots of the ensemble-averaged and spanwise-averaged liquid volume fraction are provided in appendix F to give a general sense of this spatial variation in the context of breaking waves.

As an endnote, recall the remark in § 5.1 of Part 1 that the analysis for the bubble-mass cascade resembles the monofractal analysis of Eyink (2005) for the energy cascade. The identification of a single power-law scaling in the bubble-size distribution under a given set of conditions also suggests monofractality in the bubble break-up dynamics (Turcotte 1986). For example, the $D^{-10/3}$ power-law scaling corresponds to a fractal dimension of 7/3, or that of a convoluted surface (see also Sreenivasan & Meneveau 1986; Sreenivasan, Ramshankar & Meneveau 1989; Sreenivasan 1991; Vassilicos & Hunt 1991). It appears that the act of averaging out the previously discussed large-scale fluctuations yields monofractal dynamics that provides an average sense of any underlying multifractal dynamics. This applies to both the volume-averaging procedure discussed above and the time-averaging procedure introduced in §5.1. The latter implicitly assumes that the dynamics in these small, localized and quasi-homogeneous regions are also quasi-stationary. The additional observation of quasi-self-similarity in these break-up dynamics in an intermediate range of bubble sizes, as discussed in §5.4.2, brings to mind the five-dimensional turbulent cascade in the three spatial dimensions, time and scale discussed by Cardesa, Vela-Martín & Jimenéz (2017).

Appendix B. Estimating the energy dissipation rate $\bar{\varepsilon}$

Recall from Part 1 and § 2 that the wavelength and deep-water-wave phase velocity of a characteristic wave are respectively used to estimate L and u_L in the context of breaking waves. These characteristic scales may then be used to estimate the characteristic energy dissipation rate $\bar{\epsilon} \sim u_L^3/L$, and thereafter the global Kolmogorov and Hinze scales in (2.1) and (2.2). However, some experiments (Rapp & Melville 1990; Na *et al.* 2016)

have suggested that the largest eddies and gaseous cavities have characteristic sizes and velocities that are an order of magnitude smaller. Other experiments have further suggested that most of the turbulence that originates from wave breaking initially resides in a region whose thickness is of the order of one wave height from the surface (Rapp & Melville 1990; Terray et al. 1996; Thomson et al. 2016). These have motivated the choice of other quantities for the estimation of characteristic wave scales. For example, typical surface trajectories during wave breaking may have motivated the development of a ballistic scaling by Drazen *et al.* (2008) that uses the wave height to estimate L and the corresponding ballistic velocity to estimate u_L . The corresponding ballistic time can be smaller than the wave period by an order of magnitude depending on the wave slope. The ballistic scaling was demonstrated to be adequate for single breaking events with moderate wave slope S, where $S = \sum_{i=1}^{N_m} a_i k_i$ may be interpreted as the maximum slope of the initial waveform. The waveform is assumed to be a wave packet with N_m component modes, and a_i and k_i are the initial amplitude and wavenumber, respectively, of the *i*th mode. However, it has been observed (Loewen & Melville 1991; Melville 1994; Loewen, O'Dor & Skafel 1996; Drazen et al. 2008; Deane et al. 2016a,b) that the dissipation rate eventually saturates with increasing wave slope due to the presence of multiple breaking events. This necessitates a different choice of the characteristic scales for steeper waves or more general breaking patterns. In this work, for example, the characteristic maximum slope of the initial waveform is $S = \sum_{i=1}^{3} a_i k_i = 0.76$, where i = 2 and i = 3 refer, respectively, to the first and second harmonics. This value of S is in the saturated regime identified by Drazen et al. (2008). The forthcoming discussion suggests that in the absence of additional information, the wavelength and phase velocity remain the most generic choices for the characteristic length and velocity scales, respectively, in breaking waves.

Two canonical scenarios are considered for energy conversion during wave breaking. First, consider the average energy transfer rate from coherent to turbulent motion in a single travelling wave of height h and wavelength λ . Before breaking occurs, the wave energy per unit width and wavelength $\frac{1}{4}\rho_l gh^2$ is carried forward at the group velocity $\sqrt{(g\lambda)/(8\pi)}$. Suppose that the action of breaking is to transfer this energy into a volume with cross-sectional area h by h (Lamarre & Melville 1994; Loewen & Melville 1994; Deane & Stokes 2002; Drazen et al. 2008; Rojas & Loewen 2010). The resulting average energy transfer rate per unit mass is $\bar{\varepsilon} \sim 0.8 c_p^3 / \lambda$, where c_p is the wave phase velocity. Second, consider the average energy transfer rate from large to small waves in a travelling wavepacket with central wavelength λ and a corresponding central phase velocity c_p . Gemmrich & Farmer (2004) and Babanin et al. (2010) reported wave and velocity spectra from individual wave-breaking events indicating the momentary upshifting of the peak wavenumber towards smaller scales before breaking, and then downshifting towards larger scales after breaking. This hints at the accumulation and then release of energy at some small limiting scale. A similar focusing process was observed in physical space during the prebreaking phase in the two-dimensional numerical study of a spilling breaker by Iafrati (2011), and in the experimental study of a plunging breaker by Bonmarin (1989). Inspired by these observations, as well as the arguments by Kitaigorodskii (1983), one may estimate $\bar{\varepsilon}$ as a ratio of the characteristic energy transported by the wave c_p^2 to the time scale associated with the limiting breaking scale $t_b \sim L_b/u_b$. Assuming L_b is a function of only $\bar{\varepsilon}$ and g, and taking u_b to be the phase velocity of a wave of length L_b , one eventually obtains the estimate $\bar{\varepsilon} \sim 4c_p^3/\lambda$. Both estimates for $\bar{\varepsilon}$ are of the order $O(c_p^3/\lambda)$. In the absence of additional constraints to restrict this estimate further, it is proposed that $\bar{\varepsilon} \sim u_I^3/L$ be estimated by assuming $L \sim \lambda$ and $u_L \sim c_p$.

Appendix C. Bubble identification

The error incurred by the bubble identification algorithm introduced in § 4.1 is briefly discussed for a number of simple test cases. The reader is referred to the discussion of Chan *et al.* (2018*a*) and Chan (2020) for a more detailed comparison of the employed grouping criterion with other criteria. First, the error in computing the volume of a single drop was determined by ballistically advecting the drop in a quiescent gas and by advecting it with a constant imposed velocity. In both cases, a uniform mesh was adopted, and there were O(10) grid cells spanning the drop diameter. The volume error normalized by the grid-cell volume (ΔD)³ was of the order of $O(10^{-2}-10^{-1})$, while the centroid error normalized by the grid-cell spacing ΔD was of the order of $O(10^{-4})$. Next, these errors were used to estimate the ability of the algorithm to distinguish between a closely spaced large drop–small drop pair and fluctuations in the volume of the large drop from one time step to another. The performance of the algorithm on this task is crucial in determining the accuracy of the tracking algorithm in § 4.2. It is first assumed that the normalized volume error of a drop, $\Delta V_{err}/(\Delta D)^3$, is proportional to its normalized surface area, $\pi D^2/(\Delta D)^2$:

$$\frac{\Delta \mathcal{V}_{err}}{(\Delta D)^3} = M \left[\pi \frac{D^2}{(\Delta D)^2} \right]. \tag{C1}$$

The proportionality constant, M, is assumed to depend only on the grouping criterion. Next, suppose d and D are the diameters of the small and large drops, respectively. Then, in the limit that the two scenarios cannot be distinguished, one may write

$$M\left[\pi \frac{D^2}{(\Delta D)^2}\right] \simeq \frac{\pi d^3}{6(\Delta D)^3}.$$
 (C2)

This yields the limiting size ratio

$$r = \frac{D}{d} \simeq \sqrt{\frac{d}{6M\Delta D}}.$$
 (C3)

For the grouping criterion in this work, $M = O(10^{-4})$ and $r/\sqrt{N} \simeq 20$ -40 for a given limiting bubble resolution $N = d/(\Delta D)$, so the algorithm can distinguish between these two scenarios for size ratios of the order of $O(10^2)$ even when the small bubble is not well resolved. Note that this estimate for M leads to the non-dimensional bubble diameter error for the breaking-wave baseline ensemble $\Delta D_{err}/L = O(10^{-6})$.

It is worth noting that many identification schemes are afflicted by the corner case of distinguishing two bubbles spaced a grid cell apart from a dumbbell-shaped bubble where two gaseous masses are connected by a thin gaseous bridge with the dimensions of a grid cell. The occurrence of this corner case does not necessarily suggest a deficiency in any of these schemes. Instead, it is representative of an inherent limitation of a sharp-interface field with finite resolution: when a thin gaseous bridge is numerically indistinguishable from a small under-resolved bubble or a small gaseous protrusion on the surface of a larger bubble, none of the geometries necessarily represent reality more accurately in the absence of additional information. Consequently, any decision by any scheme to favour any of the geometries is essentially arbitrary to a certain degree. The grouping criterion introduced in this work collectively identifies the gaseous masses and the gaseous bridge as a single bubble if the bridge is connected to cells in both gaseous masses with sufficiently large $1 - \phi$. Otherwise, it returns two large bubbles and a small under-resolved bubble.

Appendix D. Event detection

The constraints required to track a bubble from one simulation snapshot to another in the event detection algorithm introduced in § 4.2 are discussed here. As a bubble is advected, its mass – and volume in an incompressible setting – remains constant between the snapshots even if it deforms, within the error identified in appendix C. The principle of mass conservation is necessarily satisfied even if one bubble breaks up into two or if two bubbles coalesce into one. Suppose bubble 0 breaks up into bubbles 1 and 2, or 1 and 2 coalesce to form 0. In the case of break-up, one may write

$$\left|\mathcal{V}_{0}^{n} - \left[\mathcal{V}_{1}^{n+1} + \mathcal{V}_{2}^{n+1}\right]\right| < \Delta \mathcal{V}_{err},\tag{D1}$$

while in the case of coalescence, one may write

$$\left|\mathcal{V}_{0}^{n+1} - \left[\mathcal{V}_{1}^{n} + \mathcal{V}_{2}^{n}\right]\right| < \Delta \mathcal{V}_{err},\tag{D2}$$

where \mathcal{V}_i^j is the volume of bubble *i* in a flow snapshot *j*, *n* denotes a generic flow snapshot and $\Delta \mathcal{V}_{err}$ is the volume error discussed in appendix C. Also, if the simulation satisfies the CFL condition, then each fluid–fluid interface cannot traverse more than a single cell width every time step, multiplied by the maximum permissible Courant number *C'* for the employed advection scheme. Thus, the centroid of a bubble remains stationary between snapshots insofar as the permissible distance error is the product of the local grid spacing Δx , the number of time steps between the snapshots N_t and *C'*. This condition is also necessarily satisfied during break-up and coalescence. Denote the centroid of a bubble *i* in snapshot *j* as x_i^j . In the case of break-up, one may write

$$\left\| x_0^n - \frac{x_1^{n+1} \mathcal{V}_1^{n+1} + x_2^{n+1} \mathcal{V}_2^{n+1}}{\mathcal{V}_1^{n+1} + \mathcal{V}_2^{n+1}} \right\| < C' N_t \Delta x,$$
(D3)

while in the case of coalescence, one may write

$$\left\| \left| \mathbf{x}_{0}^{n+1} - \frac{\mathbf{x}_{1}^{n} \mathcal{V}_{1}^{n} + \mathbf{x}_{2}^{n} \mathcal{V}_{2}^{n}}{\mathcal{V}_{1}^{n} + \mathcal{V}_{2}^{n}} \right\| \right\| < C' N_{t} \Delta x.$$
(D4)

Note that the identification algorithm also generates spatial errors in the centroid. However, as noted in appendix C, these errors are typically much smaller than the grid spacing, and will be neglected. The constraints (D1)–(D4) are sufficient for the identification of break-up and coalescence events, as well as continuing bubbles, using the bubble volumes and centroids. These constraints assume that all break-up and coalescence events are binary. Note also that (D1) and (D2) imply a critical size ratio r above which events involving a small bubble and a large bubble cannot be distinguished from fluctuations in the volume of the large bubble between snapshots. This ratio is discussed in appendix C. Choosing an identification scheme with a lower volume error increases the critical size ratio keeping the resolution of the smallest bubble constant, allowing more relevant events to be captured accurately.

Appendix E. The role of coalescence

Given that the importance of coalescence events is expected to increase with the passage of time and the increase in the number density of small bubbles, coalescence statistics were investigated to determine the role of coalescence in the late wave-breaking stages.

Bubble statistics from breaking-wave simulations



Figure 16. The ratio of the size-averaged coalescence flux to the size-averaged break-up flux, $\overline{W_c}^D / \overline{W_b}^D(t)$, against non-dimensional time *t* for the 20-realization baseline ensemble subset. Coalescence events involving subgrid parent (ancestral) bubbles are excluded. For a description of the shaded regions, excluded break-up events, error bars and the bubble-size subrange over which the fluxes were averaged, refer to the caption of figure 15.

Note that coalescence events driven by large-scale dynamics such as buoyancy and turbulence are governed by the corresponding large-scale characteristic time scales (see e.g. Rodríguez-Rodríguez et al. 2006). This means that while the precise moment of coalescence may not be captured accurately since most full-scale numerical simulations cannot feasibly resolve the film drainage that precedes coalescence, macroscopic coalescence statistics remain reasonably reliable unless the system is devoid of these large-scale external influences. There is, however, a possibility that numerical coalescence elevates the total observed coalescence flux. The coalescence flux $\overline{W_c}(D; t)$ across a cutoff size D was computed in an analogous fashion to $\overline{W_h}(D; t)$ using the event detection algorithm in § 4.2. Figure 16 plots the time evolution of $\overline{W_c}^D / \overline{W_b}^D(t)$ where size averaging is performed over the same bubble-size subrange used in figure 15. The plot indicates that $\overline{W_c}^D$ is consistently smaller than $\overline{W_b}^D$, indicating that coalescence is possibly not a major player in the dynamics even towards t = 4. A closer look at the statistics suggests that the complete story may be more nuanced. Figure 17 plots the time-averaged coalescence probability $\overline{\check{q}_c}^T (D_{cp}^3 | D_{cc}^3)$, which is the probability that a coalescence event generating a child (descendant) bubble of size D_{cc} involves a parent (ancestral) bubble of size D_{cp} and another parent bubble such that the total gaseous volume remains constant, for the two time intervals identified in §§ 3.3, 5.1 and 5.2. The distributions reveal that for most child bubbles, large-size-ratio events involving one small parent bubble and one large parent bubble dominate the dynamics. The dominance of large-size-ratio coalescence events may be analytically supported from kinetic theory arguments, such as the one posited by Chan et al. (2018b), which argues that large-size-ratio collisions are favoured when the size distribution decays with increasing bubble size. In short, the dynamics of coalescence appears to be dominated by parent bubbles that are beyond the reach of the current simulations. It is possible, then, that $\overline{W_c}$ has been underestimated in these simulations, and the jury is still out as to the importance of coalescence in the late wave-breaking stages. A study of coalescence is deferred to later work where the capability to computationally capture these small bubbles is included. One possible approach is to



W.H.R. Chan, P.L. Johnson, P. Moin and J. Urzay

Figure 17. The time-averaged coalescence probability $\overline{\dot{q}_c}^T (D_{cp,j}^3 | D_{cc}^3)$ against normalized parent (ancestral) bubble volume D_{cp}^3/D_{cc}^3 for the 20-realization baseline ensemble subset during the (a,c,e) early (t = 2.30-3.14) and (b,d,f) late (t = 3.45-3.95) wave-breaking stages. The non-dimensional child (descendant) bubble sizes considered are (a,b) between 1.86×10^{-2} and 2.23×10^{-2} , (c,d) between 2.68×10^{-2} and 3.23×10^{-2} and (e,f) between 3.54×10^{-2} and 4.25×10^{-2} . Only parent bubble volumes in the coalescence event where one of the parent bubble radii corresponds to this resolution limit. The error bars denote one standard error over the baseline ensemble.

replace bubbles under-resolved by the current Eulerian treatment with Lagrangian point particles, as alluded to in Part 1 and to be discussed in forthcoming work.

Appendix F. Spatial distribution of the liquid volume fraction

In order to provide a general sense of how the bubbles are spatially distributed during the wave-breaking process, figure 18 plots the ensemble-averaged and spanwise-averaged liquid volume fraction, ϕ , at two representative time instances, one just before the first



Figure 18. The ensemble-averaged and spanwise-averaged liquid volume fraction ϕ for the 30-realization baseline ensemble, computed at (a) t = 2.23 and (b) t = 4.04. The solid line denotes the $\phi = 0.5$ isocontour.

characteristic time interval and one just after the second characteristic time interval identified in table 2. Note that the void fraction may be correspondingly computed as $(1 - \phi)$. In general agreement with the other works discussed in §§ 3.3 and 5.1, high-void-fraction mixed-phase regions are initially concentrated near the areas with rapid topology change as evidenced in figure 18(*a*), and then gradually evolve into a number of diffuse bubble clouds near the wave surface as evidenced in figure 18(*b*).

REFERENCES

- AGRAWAL, Y.C., TERRAY, E.A., DONELAN, M.A., HWANG, P.A., WILLIAMS, A.J. III, DRENNAN, W.M., KAHMA, K.K. & KITAIGORODSKII, S.A. 1992 Enhanced dissipation of kinetic energy beneath surface waves. *Nature* 359, 219–220.
- ATKINSON, L.P. 1973 Effect of air bubble solution on air-sea gas exchange. J. Geophys. Res. 78 (6), 962–968.
- BABANIN, A.V., CHALIKOV, D., YOUNG, I.R. & SAVELYEV, I. 2010 Numerical and laboratory investigation of breaking of steep two-dimensional waves in deep water. J. Fluid Mech. 644, 433–463.
- BLANCHARD, D.C. & SYZDEK, L. 1970 Mechanism for the water-to-air transfer and concentration of bacteria. Science 170 (3958), 626–628.
- BLANCHARD, D.C. & WOODCOCK, A.H. 1957 Bubble formation and modification in the sea and its meteorological significance. *Tellus* 9, 145–158.
- BLENKINSOPP, C.E. & CHAPLIN, J.R. 2007 Void fraction measurements in breaking waves. *Proc. R. Soc. Lond.* A 463, 3151–3170.
- BLENKINSOPP, C.E. & CHAPLIN, J.R. 2010 Bubble size measurements in breaking waves using optical fiber phase detection probes. *IEEE J. Ocean Engng* **35**, 388–401.
- BONMARIN, P. 1989 Geometric properties of deep-water breaking waves. J. Fluid Mech. 209, 405-433.
- BRAVO, L., KIM, D., HAM, F. & KERNER, K. 2018a High fidelity simulations of primary breakup and vaporization of liquid jet in cross flow. AIAA Paper 2018-4683.
- BRAVO, L., KIM, D., HAM, F., MATUSIK, K.E., DUKE, D.J., KASTENGREN, A.L., SWANTEK, A.B. & POWELL, C.F. 2016 Numerical investigation of liquid jet breakup and droplet statistics with comparison to X-ray radiography. *AIAA Paper* 2016-5096.
- BRAVO, L., KIM, D., HAM, F., POWELL, C. & KASTENGREN, A. 2019 Effects of fuel viscosity on the primary breakup dynamics of a high-speed liquid jet with comparison to X-ray radiography. *Proc. Combust. Inst.* 37, 3245–3253.
- BRAVO, L., KIM, D., HAM, F. & SU, S. 2018b Computational study of atomization and fuel drop size distributions in high-speed primary breakup. Atomiz. Sprays 28 (4), 321–344.
- BRAVO, L., KIM, D., TESS, M., KURMAN, M., HAM, F. & KWEON, C. 2015 High resolution numerical simulations of primary atomization in diesel sprays with single component reference fuels. In *Proceedings* of the ILASS Americas 27th Annual Conference on Liquid Atomization and Spray Systems. Institute for Liquid Atomization and Spray Systems, North and South America.

- CARDESA, J.I., VELA-MARTÍN, A. & JIMENÉZ, J. 2017 The turbulent cascade in five dimensions. Science 357, 782–784.
- CASTRO, A.M., LI, J. & CARRICA, P.M. 2016 A mechanistic model of bubble entrainment in turbulent free surface flows. Intl J. Multiphase Flow 86, 35–55.
- CHAN, W.H.R. 2020 The bubble breakup cascade in turbulent breaking waves and its implications on subgrid-scale modeling. PhD thesis, Stanford University.
- CHAN, W.H.R., DODD, M.S., JOHNSON, P.L., URZAY, J. & MOIN, P. 2018a Formation and dynamics of bubbles generated in breaking waves: Part I. Algorithms for the identification of bubbles and breakup/coalescence events. In *Center for Turbulence Research Annual Research Briefs*, pp. 3–20. Stanford University.
- CHAN, W.H.R., DODD, M.S., JOHNSON, P.L., URZAY, J. & MOIN, P. 2018b Formation and dynamics of bubbles generated in breaking waves: Part II. The evolution of the bubble size distribution and breakup/coalescence statistics. In *Center for Turbulence Research Annual Research Briefs*, pp. 21–34. Stanford University.
- CHAN, W.H.R., JOHNSON, P.L. & MOIN, P. 2021 The turbulent bubble break-up cascade. Part 1. Theoretical developments. J. Fluid Mech. 912, A42.
- CHAN, W.H.R., MIRJALILI, S., JAIN, S.S., URZAY, J., MANI, A. & MOIN, P. 2019 Birth of microbubbles in turbulent breaking waves. *Phys. Rev. Fluids* **4**, 100508.
- CHAN, W.H.R., URZAY, J. & MOIN, P. 2018c Subgrid-scale modeling for microbubble generation amid colliding water surfaces. In *Proceedings of the 32nd Symposium on Naval Hydrodynamics*. U.S. Office of Naval Research (ONR) and the Institute for Fluid Dynamics and Ship Theory of the Hamburg University of Technology (TUHH).
- CHEN, G., KHARIF, C., ZALESKI, S. & LI, J. 1999 Two-dimensional Navier–Stokes simulation of breaking waves. *Phys. Fluids* **11** (1), 121–133.
- CROOK, J.A., JACKSON, L.S. & FORSTER, P.M. 2016 Can increasing albedo of existing ship wakes reduce climate change? J. Geophys. Res. Atmos. 121, 1549–1558.
- CUMMINS, S.J., FRANCOIS, M.M. & KOTHE, D.B. 2005 Estimating curvature from volume fractions. *Comput. Struct.* 83, 425–434.
- CZERSKI, H. 2017 Behold the bubbly ocean. Phys. World 30 (11), 34–38.
- CZERSKI, H. & DEANE, G.B. 2010 Contributions to the acoustic excitation of bubbles released from a nozzle. *J. Acoust. Soc. Am.* **128** (5), 2625–2634.
- DEANE, G.B. 1997 Sound generation and air entrainment by breaking waves in the surf zone. J. Acoust. Soc. Am. 102, 2671–2689.
- DEANE, G.B. 2016 The performance of high-frequency Doppler sonars in actively breaking wave crests. *IEEE J. Ocean Engng* **41** (4), 1028–1034.
- DEANE, G.B., STOKES, D. & CALLAGHAN, A.H. 2016a Turbulence in breaking waves. *Phys. Today* 69, 86–87.
- DEANE, G.B. & STOKES, M.D. 2002 Scale dependence of bubble creation mechanisms in breaking waves. *Nature* **418**, 839–844.
- DEANE, G.B., STOKES, M.D. & CALLAGHAN, A.H. 2016b The saturation of fluid turbulence in breaking laboratory waves and implications for whitecaps. J. Phys. Oceanogr. 46, 975–992.
- DEIKE, L., MELVILLE, W.K. & POPINET, S. 2016 Air entrainment and bubble statistics in breaking waves. *J. Fluid Mech.* **801**, 91–129.
- DEIKE, L., POPINET, S. & MELVILLE, W.K. 2015 Capillary effects on wave breaking. J. Fluid Mech. **769**, 541–569.
- DODD, M.S. & JOFRE, L. 2019 Small-scale flow topologies in decaying isotropic turbulence laden with finite-size droplets. *Phys. Rev. Fluids* **4**, 064303.
- DRAZEN, D.A., MELVILLE, W.K. & LENAIN, L. 2008 Inertial scaling of dissipation in unsteady breaking waves. J. Fluid Mech. 611, 307–332.
- DUINEVELD, P.C. 1998 Bouncing and coalescence of bubble pairs rising at high Reynolds number in pure water or aqueous surfactant solutions. In *In Fascination of Fluid Dynamics*, 1st edn (ed. A. Biesheuvel & G.J.F. van Heijst), pp. 409–439. Kluwer Academic Publishers.
- EMERSON, S. & BUSHINSKY, S. 2016 The role of bubbles during air-sea gas exchange. J. Geophys. Res. Oceans 121, 4360–4376.
- EYINK, G.L. 2005 Locality of turbulent cascades. *Physica* D 207, 91–116.
- FARMER, D.M., DEANE, G.B. & VAGLE, S. 2001 The influence of bubble clouds on acoustic propagation in the surf zone. *IEEE J. Ocean Engng* 26, 113–124.
- FARMER, D.M. & LEMON, D.D. 1984 The influence of bubbles on ambient noise in the ocean at high wind speeds. J. Phys. Oceanogr. 14 (11), 1762–1778.

- FILIPPOV, A.F. 1961 On the distribution of the sizes of particles which undergo splitting. *Theory Prob. Applics.* 6, 275–294.
- FOX, F.E. & HERZFELD, K.F. 1954 Gas bubbles with organic skin as cavitation nuclei. J. Acoust. Soc. Am. 26 (6), 984–989.
- FRANCOIS, M.M., CUMMINS, S.J., DENDY, E.D., KOTHE, D.B., SICILIAN, J.M. & WILLIAMS, M.W. 2006 A balanced-force algorithm for continuous and sharp interfacial surface tension models within a volume tracking framework. J. Comput. Phys. 213, 141–173.
- GARRETT, C., LI, M. & FARMER, D. 2000 The connection between bubble size spectra and energy dissipation rates in the upper ocean. J. Phys. Oceanogr. 30, 2163–2171.
- GARRETTSON, G.A. 1973 Bubble transport theory with application to the upper ocean. J. Fluid Mech. 59, 187–206.
- GEMMRICH, J.R. 2010 Strong turbulence in the wave crest region. J. Phys. Oceanogr. 40, 583-595.
- GEMMRICH, J.R. & FARMER, D.M. 2004 Near-surface turbulence in the presence of breaking waves. J. Phys. Oceanogr. 34, 1067–1086.
- HALL, M.V. 1989 A comprehensive model of wind-generated bubbles in the ocean and predictions of the effects on sound propagation at frequencies up to 40 kHz. J. Acoust. Soc. Am. 86, 1103–1117.
- HAM, F., KIM, D., BOSE, S., LE, H. & HERRMANN, M. 2014 Simulation of liquid fuel atomization by a complex high-shear swirling injector. In *Proceedings of ASME Turbo Expo 2014: Turbine Technical Conference and Exposition.* The American Society of Mechanical Engineers.
- HEBERT, D.A., SCHMIDT, D.P., KNAUS, D.A., PHILLIPS, S. & MAGARI, P.J. 2008 Parallel VOF spray drop identification in an unstructured grid. In *Proceedings of the ILASS Americas 21st Annual Conference on Liquid Atomization and Spray Systems*. Institute for Liquid Atomization and Spray Systems, North and South America.
- HERRMANN, M. 2008 A balanced force refined level set grid method for two-phase flows on unstructured flow solver grids. J. Comput. Phys. 227, 2674–2706.
- HERRMANN, M. 2010 A parallel Eulerian interface tracking/Lagrangian point particle multi-scale coupling procedure. J. Comput. Phys. 229, 745–759.
- HINZE, J.O. 1955 Fundamentals of the hydrodynamic mechanism of splitting in dispersion processes. AIChE J. 1, 289–295.
- IAFRATI, A. 2009 Numerical study of the effects of the breaking intensity on wave breaking flows. J. Fluid Mech. 622, 371–411.
- IAFRATI, A. 2011 Energy dissipation mechanisms in wave breaking processes: spilling and highly aerated plunging breaking events. *J. Geophys. Res. Oceans* **116**, C07024.
- IVEY, C.B. & MOIN, P. 2017 Conservative and bounded volume-of-fluid advection on unstructured grids. J. Comput. Phys. 350, 387–419.
- JOHNSON, B.D. & COOKE, R.C. 1979 Bubble populations and spectra in coastal water: a photographic approach. J. Geophys. Res. Oceans 84 (C7), 3761–3766.
- JOHNSON, B.D. & COOKE, R.C. 1980 Organic particle and aggregate formation resulting from the dissolution of bubbles in seawater. *Limnol. Oceanogr.* **25** (4), 653–661.
- JOHNSON, B.D. & WANGERSKY, P.J. 1987 Microbubbles: stabilization by monolayers of adsorbed particles. J. Geophys. Res. Oceans 92 (C13), 14641–14647.
- KANWISHER, J. 1963 Effect of wind on CO₂ exchange across the sea surface. J. Geophys. Res. 68 (13), 3921–3927.
- KIGER, K.T. & DUNCAN, J.H. 2012 Air-entrainment mechanisms in plunging jets and breaking waves. Annu. Rev. Fluid Mech. 44, 563–596.
- KIM, D., HAM, F., BOSE, S., LE, H., HERRMANN, M., LI, X., SOTERIOU, M.C. & KIM, W. 2014 High-fidelity simulation of atomization in a gas turbine injector high shear nozzle. In *Proceedings of the ILASS Americas 26th Annual Conference on Liquid Atomization and Spray Systems*. Institute for Liquid Atomization and Spray Systems, North and South America.
- KITAIGORODSKII, S.A. 1983 On the theory of the equilibrium range in the spectrum of wind-generated gravity waves. J. Phys. Oceanogr. 13, 816–827.
- KITAIGORODSKII, S.A., DONELAN, M.A., LUMLEY, J.L. & TERRAY, E.A. 1983 Wave-turbulence interactions in the upper ocean. Part II: statistical characteristics of wave and turbulent components of the random velocity field in the marine surface layer. *J. Phys. Oceanogr.* **13**, 1988–1999.
- KOLMOGOROV, A.N. 1941 The local structure of turbulence in incompressible viscous fluid for very large Reynolds numbers. *Dokl. Akad. Nauk SSSR* **30**, 299–303.
- KOLMOGOROV, A.N. 1949 On the breakage of drops in a turbulent flow. Dokl. Akad. Nauk SSSR 66, 825-828.
- KOLMOGOROV, A.N. 1962 A refinement of previous hypotheses concerning the local structure of turublence in a viscous incompressible fluid at high Reynolds number. J. Fluid Mech. 13, 82–85.

- LAMARRE, E. & MELVILLE, W.K. 1994 Void-fraction measurements and sound-speed fields in bubble plumes generated by breaking waves. J. Acoust. Soc. Am. 95, 1317–1328.
- LEHR, F., MILLIES, M. & MEWES, D. 2002 Bubble-size distributions and flow fields in bubble columns. *AIChE J.* **48**, 2426–2443.
- LIM, H.-J., CHANG, K.-A., HUANG, Z.-C. & NA, B. 2015 Experimental study on plunging breaking waves in deep water. J. Geophys. Res. Oceans 120, 2007–2049.
- LOEWEN, M.R. & MELVILLE, W.K. 1991 Microwave backscatter and acoustic radiation from breaking waves. *J. Fluid Mech.* **224**, 601–623.
- LOEWEN, M.R. & MELVILLE, W.K. 1994 An experimental investigation of the collective oscillations of bubble plumes entrained by breaking waves. J. Acoust. Soc. Am. 95, 1329–1343.
- LOEWEN, M.R., O'DOR, M.A. & SKAFEL, M.G. 1996 Bubbles generated by mechanically generated breaking waves. J. Geophys. Res. Oceans 101 (C9), 20759–20769.
- MARTÍNEZ-BAZÁN, C., MONTAÑÉS, J.L. & LASHERAS, J.C. 1999 On the breakup of an air bubble injected into a fully developed turbulent flow. Part 1. Breakup frequency. J. Fluid Mech. 401, 157–182.
- MASNADI, N., ERININ, M.A., WASHUTA, N., NASIRI, F., BALARAS, E. & DUNCAN, J.H. 2020 Air entrainment and surface fluctuations in a turbulent ship hull boundary layer. J. Ship Res. 64 (2), 185–201.
- MEDWIN, H. 1970 In situ acoustic measurements of bubble populations in coastal ocean waters. J. Geophys. Res. 75 (3), 599–611.
- MELVILLE, W.K. 1994 Energy dissipation by breaking waves. J. Phys. Oceanogr. 24, 2041–2049.
- MELVILLE, W.K. 1996 The role of surface-wave breaking in air-sea interaction. *Annu. Rev. Fluid Mech.* 28, 279–321.
- MIRJALILI, S., CHAN, W.H.R. & MANI, A. 2018 High fidelity simulations of micro-bubble shedding from retracting thin gas films in the context of liquid-liquid impact. In *Proceedings of the 32nd Symposium on Naval Hydrodynamics*. U.S. Office of Naval Research (ONR) and the Institute for Fluid Dynamics and Ship Theory of the Hamburg University of Technology (TUHH).
- MIRJALILI, S., IVEY, C.B. & MANI, A. 2019 Comparison between the diffuse interface and volume of fluid methods for simulating two-phase flows. *Intl J. Multiphase Flow* 116, 221–238.
- MIRJALILI, S. & MANI, A. 2020 Transitional stages of thin air film entrapment in drop-pool impact events. *J. Fluid Mech.* **901**, A14.
- MORTAZAVI, M. 2016 Air entrainment and micro-bubble generation by turbulent breaking waves. PhD thesis, Stanford University.
- MORTAZAVI, M., LE CHENADEC, V., MOIN, P. & MANI, A. 2016 Direct numerical simulation of a turbulent hydraulic jump: turbulence statistics and air entrainment. *J. Fluid Mech.* **797**, 60–94.
- NA, B., CHANG, K.-A., HUANG, Z.-C. & LIM, H.-J. 2016 Turbulent flow field and air entrainment in laboratory plunging breaking waves. J. Geophys. Res. Oceans 121, 2980–3009.
- ONSAGER, L. 1945 The distribution of energy in turbulence. Phys. Rev. 68 (11-12), 286.
- POPE, S.B. 2000 Turbulent Flows. Cambridge University Press.
- PROSPERETTI, A. 1988 Bubble dynamics in oceanic ambient noise. In Sea Surface Sound: Natural Mechanisms of Surface-Generated Noise in the Ocean, 1st edn (ed. B.R. Kerman), pp. 151–171. Kluwer Academic Publishers.
- QI, Y., MASUK, A.U.M. & NI, R. 2020 Towards a model of bubble breakup in turbulence through experimental constraints. *Intl J. Multiphase Flow* 132, 103397.
- RAPP, R.J. & MELVILLE, W.K. 1990 Laboratory measurements of deep-water breaking waves. *Phil. Trans. R. Soc. Lond.* A 331, 735–808.
- REED, A.M. & MILGRAM, J.H. 2002 Ship wakes and their radar images. *Annu. Rev. Fluid Mech.* 34, 469–502.
- RICHARDSON, L.F. 1922 Weather Prediction by Numerical Process. Cambridge University Press.
- RODRÍGUEZ-RODRÍGUEZ, J., GORDILLO, J.M. & MARTÍNEZ-BAZÁN, C. 2006 Breakup time and morphology of drops and bubbles in a high-Reynolds-number flow. *J. Fluid Mech.* **548**, 69–86.
- ROJAS, G. & LOEWEN, M.R. 2010 Void fraction measurements beneath plunging and spilling breaking waves. J. Geophys. Res. Oceans 115, C08001.
- RUBEL, C. & OWKES, M. 2019 Extraction of droplet genealogies from high-fidelity atomization strategies. *Atomiz. Sprays* 29, 709–739.
- SCARDOVELLI, R. & ZALESKI, S. 1999 Direct numerical simulation of free-surface and interfacial flow. *Annu. Rev. Fluid Mech.* **31**, 567–603.
- SCHULKIN, M. 1968 Surface-coupled losses in surface sound channels. J. Acoust. Soc. Am. 44, 1152–1154.
- SCHULKIN, M. 1969 Surface-coupled losses in surface sound channel propagation. II. J. Acoust. Soc. Am. 44, 1152–1154.

SEITZ, R. 2011 Bright water: hydrosols, water conservation and climate change. J. Clim. 105, 365-381.

SREENIVASAN, K.R. 1991 Fractals and multifractals in fluid turbulence. Annu. Rev. Fluid Mech. 23, 539-600.

SREENIVASAN, K.R. & MENEVEAU, C. 1986 The fractal facets of turbulence. J. Fluid Mech. 173, 357–386.

- SREENIVASAN, K.R., RAMSHANKAR, R. & MENEVEAU, C. 1989 Mixing, entrainment and fractal dimensions of surfaces in turbulent flows. Proc. R. Soc. Lond. A 421, 79–108.
- STANIC, S., CARUTHERS, J.W., GOODMAN, R.R., KENNEDY, E. & BROWN, R.A. 2009 Attenuation measurements across surface-ship wakes and computed bubble distributions and void fractions. *IEEE J. Ocean Engng* 34, 83–92.
- STEFAN, R.L. & SZERI, A.J. 1999 Surfactant scavenging and surface deposition by rising bubbles. J. Colloid Interface Sci. 212, 1–13.
- STRAMSKI, D. 1994 Gas microbubbles: an assessment of their significance to light scattering in quiescent seas. *Proc. SPIE* **2258**, 704–710.
- STRAMSKI, D., BOSS, E., BOGUCKI, D. & VOSS, K.J. 2004 The role of seawater constituents in light backscattering in the ocean. *Prog. Oceanogr.* 61 (1), 27–56.
- STRASBERG, M. 1956 Gas bubbles as sources of sound in liquids. J. Acoust. Soc. Am. 28 (1), 20-26.
- TAKAGI, S. & MATSUMOTO, Y. 2011 Surfactant effects on bubble motion and bubbly flows. *Annu. Rev. Fluid Mech.* 43, 615–636.
- TAVAKOLINEJAD, M. 2010 Air bubble entrainment by breaking bow waves simulated by a 2D + T technique. PhD thesis, University of Maryland, College Park.
- TERRAY, E.A., DONELAN, M.A., AGRAWAL, Y.C., DRENNAN, W.M., KAHMA, K.K., WILLIAMS, A.J. III, HWANG, P.A. & KITAIGORODSKII, S.A. 1996 Estimates of kinetic energy dissipation under breaking waves. J. Phys. Oceanogr. 26, 792–807.
- TERRILL, E.J., MELVILLE, W.K. & STRAMSKI, D. 2001 Bubble entrainment by breaking waves and their influence on optical scattering in the upper ocean. J. Geophys. Res. Oceans 106 (C8), 16815–16823.
- THOMSON, J., SCHWENDEMAN, M.S., ZIPPEL, S.F., MOGHIMI, S., GEMMRICH, J. & ROGERS, W.E. 2016 Wave-breaking turbulence in the ocean surface layer. *J. Phys. Oceanogr.* **46**, 1857–1870.
- THORPE, S.A. 1982 On the clouds of bubbles formed by breaking wind-waves in deep water, and their role in air-sea gas transfer. *Phil. Trans. R. Soc. Lond.* A **304**, 155–210.
- THORPE, S.A. 1992 Bubble clouds and the dynamics of the upper ocean. Q. J. R. Meteorol. Soc. 118, 1–22.
- THORPE, S.A. 1995 Dynamical processes of transfer at the sea surface. Prog. Oceanogr. 35, 315–352.
- TOMAR, G., FUSTER, D., ZALESKI, S. & POPINET, S. 2010 Multiscale simulations of primary atomization. *Comput. Fluids* 39, 1864–1874.
- TREVORROW, M.V., VAGLE, S. & FARMER, D.M. 1994 Acoustical measurements of microbubbles within ship wakes. J. Acoust. Soc. Am. 95, 1922–1930.
- TURCOTTE, D.L. 1986 Fractals and fragmentation. J. Geophys. Res. Solid Earth 91, 1921–1926.
- TURNER, W.R. 1961 Microbubble persistence in fresh water. J. Acoust. Soc. Am. 33 (9), 1223–1233.
- TWARDOWSKI, M., ZHANG, X., VAGLE, S., SULLIVAN, J., FREEMAN, S., CZERSKI, H., YOU, Y., BI, L. & KATTAWAR, G. 2012 The optical volume scattering function in a surf zone inverted to derive sediment and bubble particle subpopulations. J. Geophys. Res. Oceans 117, C00H17.
- VAGLE, S. & FARMER, D.M. 1998 A comparison of four methods for bubble size and void fraction measurements. *IEEE J. Ocean Engng* 23 (3), 211–222.
- VASSILICOS, J.C. & HUNT, J.C.R. 1991 Fractal dimensions and spectra of interfaces with application to turbulence. Proc. R. Soc. Lond. A 435, 505–534.
- VERON, F. 2015 Ocean spray. Annu. Rev. Fluid Mech. 47, 507-538.
- WANG, Z., YANG, J. & STERN, F. 2016 High-fidelity simulations of bubble, droplet and spray formation in breaking waves. J. Fluid Mech. 792, 307–327.
- WANNINKHOF, R., ASHER, W.E., HO, D.T., SWEENEY, C. & MCGILLIS, W.R. 2009 Advances in quantifying air-sea gas exchange and environmental forcing. *Annu. Rev. Mar. Sci.* **1**, 213–244.
- WEBER, M.E., BLANCHARD, D.C. & SYZDEK, L. 1983 The mechanism of scavenging of waterborne bacteria by a rising bubble. *Limnol. Oceanogr.* 28 (1), 101–105.
- WOOLF, D.K. 1993 Bubbles and the air-sea transfer velocity of gases. Atmos. Ocean 31 (4), 517-540.
- YU, X., HENDRICKSON, K., CAMPBELL, B.K. & YUE, D.K.P. 2019 Numerical investigation of shear-flow free-surface turbulence and air entrainment at large froude and weber numbers. *J. Fluid Mech.* **880**, 209–238.
- YU, X., HENDRICKSON, K. & YUE, D.K.P. 2020 Scale separation and dependence of entrainment bubble-size distribution in free-surface turbulence. J. Fluid Mech. 885, R2.
- ZHANG, X., LEWIS, M., BISSETT, W.P., JOHNSON, B. & KOHLER, D. 2004 Optical influence of ship wakes. *Appl. Opt.* **43**, 3122–3132.
- ZHANG, X., LEWIS, M. & JOHNSON, B. 1998 Influence of bubbles on scattering of light in the ocean. Appl. Opt. 37, 6525–6536.